



Article The Medical Right to Repair: Intellectual Property, the Maker Movement, and COVID-19

Matthew Rimmer

check for **updates**

Citation: Rimmer, M. The Medical Right to Repair: Intellectual Property, the Maker Movement, and COVID-19. *Sustainability* **2023**, *15*, 14839. https://doi.org/10.3390/ su152014839

Academic Editor: Michela Mortara

Received: 31 July 2023 Revised: 6 September 2023 Accepted: 13 September 2023 Published: 13 October 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Faculty of Business and Law, Queensland University of Technology (QUT), Brisbane, QLD 4000, Australia; matthew.rimmer@qut.edu.au

Abstract: This article considers the strengths and limitations of the use of 3D printing and additive manufacturing for the production of personal protective equipment (PPE) during the COVID-19 public health crisis. It explores the role of the Maker Movement in addressing the shortfall in medical equipment during the public health pandemic. Taking a comparative approach, this article evaluates the responses of both the commercial and amateur sides of the 3D printing community to the COVID-19 public health crisis. In the EU, the Fab Lab Network sought to overcome a breakdown in supply chains. In the US, Dale Dougherty of Make Magazine promoted Plan C, in which volunteers have worked together to produce PPE. In Australia, 3D printing has been used to augment supplies of PPE. In this context, a key issue across jurisdictions has been the tension between intellectual property rights and the right to repair—particularly when 3D printing has been applied to deal with shortages in PPE. Senator Ron Wyden put forward the Critical Medical Infrastructure Right-to-Repair Act of 2020 (US) to try to resolve these tensions. Open licensing has proven to be a helpful mechanism to enable open collaboration and sharing of 3D printing designs for the purposes of health care. Nonetheless, it is argued that there should be stronger recognition of the right to repair-particularly in the context of health-care and medical devices. The COVID-19 crisis has highlighted that there needs to be a broader array of intellectual property flexibilities to deal with public health emergencies—including in respect of the right to repair. As such, this article supports a broad vision of a TRIPS Waiver which includes the right to repair. The recognition of a right to repair will help promote the realization of the Sustainable Development Goals and a COVID-19 recovery. The topic of the medical right to repair has larger implications for sustainability and the Sustainable Development Goals (SDGs), particularly in respect of responsible production and consumption (SDG 12), as well as good health and well-being (SDG 3), innovation (SDG 9), and partnerships for the goals (SDG 17).

Keywords: intellectual property; 3D printing; the Maker Movement; the right to repair; COVID-19; sustainability; sustainable development goals; pandemics

1. Introduction

During the COVID-19 crisis, a catch cry of doctors, nurses, and health professionals was 'Get Me PPE' [1]. There was great concern amongst front-line health workers about shortages of personal protective equipment (PPE) (such as face masks, respirators, and other supplies) during the COVID-19 public health emergency.

The World Health Organization (WHO) called for a boost to domestic production of PPE by governments across the world [2]. WHO director-general Dr. Tedros Adhanom Ghebreyesus commented: 'Without secure supply chains, the risk to healthcare workers around the world is real' [2]. He observed: 'Industry and governments must act quickly to boost supply, ease export restrictions and put measures in place to stop speculation and hoarding' [2]. The WHO director-general emphasized: 'We can't stop COVID-19 without protecting health workers first' [2].

In this context, there was much interest in the application of 3D printing and additive manufacturing to address supply shortages—particularly in respect of PPE and other

medical supplies and specialized products [3]. In particular, the Maker Movement has sought to provide help and assistance to public health authorities through 3D printing PPE and medical supplies.

The anthropologist Dr. Sally Applin comments that the Maker Movement's can-do DIY spirit has been helpful in response to the COVID-19 pandemic [4]. She provides an overview of the response of the Maker Movement: 'Right now, all over the world, people are engaging in the construction of critical supplies and [PPE] respirators, and other clinical items, using tools and techniques such as 3D printing, sewing, repurposing factory equipment, and other skills and ways of thinking' [4]. Applin discusses the origin and the evolution of the Maker Movement, and details a number of 3D printing projects in the COVID-19 era. She observed that the education and activity networks of the Maker Movement 'have created a pipeline for information of how to innovate PPEs rapidly across the globe' [4]. She commented: 'It is this history that now critically comes into play with the coronavirus, for the spirit of hacking, making, and co-operative learning going on around the world is based on a straight-out-of-Maker Faire ethos' [4].

Nonetheless, the adoption of 3D printing of PPE and medical supplies raises larger questions about the operation of intellectual property-particularly in respect of the right to repair. Cory Doctorow has contemplated the right to repair in the times of the COVID-19 pandemic [5]. He considered the local 3D printing of replacement parts for ventilators in a hospital in Brescia, Italy, and the ensuing controversy over intellectual property and the right to repair. Doctorow highlighted how the COVID-19 crisis disrupted trade and logistics: 'The global supply-chain shutdown has revealed the fragility of long distance, complex manufacturing systems that are organized around central hubs that represent points of critical failure' [5]. He welcomed the response of the Maker Movement to the COVID-19 crisis: 'The surge in open source hardware designs and parts for medical equipment during the emergency represents a distributed, urgently needed decentralization of our world's critical manufacturing capacity' [5]. Doctorow emphasized: 'The right person to decide whether a field repair should be attempted, and whether the repair is solid enough to rely upon are medical professionals, not the shareholders of med-tech companies or the lawyers who write their terms of service and patent applications' [5]. Doctorow maintained: 'Today, we need those companies to step up by providing repair instructions, specifications, and technical aid to the global volunteer corps of makers and fixers who have given themselves over to helping us all weather this calamity' [5]. Doctorow has expressed broader concerns about the development of Big Tech monopolies through the abuse of intellectual property rights [6].

This article provides an evaluation of intellectual property and the right to repair in light of the public health response to the COVID-19 crisis. It builds upon the literature in respect of several domains. This article contributes to the field of intellectual property and public health—particularly as it concerns access to medicines, and other essential medical technologies during public health emergencies [7–13]. This article makes a contribution to the literature on the right to repair—particularly focusing upon the applications in relation to health-care and medicine [14–17]. This research is also part of a larger body of work on 3D printing regulation—particularly in relation to intellectual property law and policy [18–24]. This scholarship is also part of the emerging and growing field of COVID-19 law and regulation [25–30]. This article also contributes to the growing body of work on intellectual property and sustainable development—focusing, as it does, on the key area of the right to repair, which is designed to promote responsible and sustainable consumption in a circular economy [31–34].

In terms of its methodology, this article is not a traditional piece of black letter legal analysis. There is yet to be clear cut case law on the matters raised in this field. The topic of 3D printing and the COVID-19 crisis demands a mixture of methodologies to make sense of the subject matter. This article provides a history of the community and industry-based 3D printing responses in the COVID-19 crisis. It particularly highlights some of the innovation policies and community responses in respect of 3D printing PPE and medical equipment.

This article also explores the debate over the right to repair, with a particular consideration of medical and health matters during the COVID-19 crisis. It argues that there needs to be further reform of intellectual property laws—particularly in respect to the right to repair and health care. This article also touches upon larger questions around intellectual property and international trade. In particular, the question of a medical right to repair comes within the ambit of the *TRIPS Waiver* [35,36] and the *Ministerial Decision* [37,38]. The topic is part of a larger debate about what intellectual property exceptions should be available during a public health emergency—such as the COVID-19 crisis. It is certainly acknowledged that 3D printing and additive manufacturing of medical supplies during the COVID-19 crisis also raises other policy issues around medical regulation and product liability. However, a full analysis of those adjacent issues is beyond the scope of this particular paper.

This article is a comparative piece of work, looking at intellectual property policy, law, and practice. It compares the legal and political and public health responses of various key jurisdictions—including the European Union (EU), the United States, and Australia. Part 2 focuses on the EU and its approach to intellectual property, 3D printing, and the right to repair during the COVID-19 crisis. It considers the controversy over the 3D printing of replacement valves for ventilators in Italian hospitals during the COVID-19 crisis. Part 3 considers the situation of the US in respect of the right to repair during the COVID-19 emergency. In particular, it examines the Critical Medical Infrastructure Right-to-Repair Act of 2020 (US), as well as more general proposals for a right to repair. Part 4 explores the use of 3D printing by Australian universities and industry to provide PPE supplies during the COVID-19 crisis. It considers the Productivity Commission inquiry into the right to repair, and its implications for medicine and healthcare. Part 5 explores the scope for the right to repair in the field of medical matters under international law—particularly the TRIPS Agreement 1994. It explores whether the proposed TRIPS Waiver and Ministerial Decision should have been framed broadly enough to include the right to repair. The conclusion recommends that there should be law reform to provide recognition of the right to repair—with specific reference to the repair of medical supplies during public health emergencies. It maintains that the full recognition of the right to repair will promote the UN Sustainable Development Goals (SDGs)—in particular, promoting responsible production and consumption (SDG 12), as well as good health and well-being (SDG 3), innovation (SDG 9), and partnerships for the goals (SDG 17).

2. The European Union

The European Union (EU) is notable for its strengths in respect of additive manufacturing and industrial 3D printing [39]. In particular, Germany, the Netherlands, and Spain are regarded as leaders in 3D printing. The European Patent Office has undertaken data analysis of the intellectual property landscapes in respect of 3D printing [40]. The field of 3D printing patents is becoming quite crowded, with a thicket of patents. The European Commission has also published a commissioned report on the intellectual property implications of the development of industrial 3D printing in 2020 [41].

In addition to the commercial sector of 3D printing, there is a thriving Maker Movement in the EU. Following the example of the Chaos Computer Club, hackerspaces have a long tradition and standing in the EU [42]. The Fab Lab movement has been particularly popular in the EU. Networks of Fab Labs have collaborated with one another on a range of projects [43,44]. There are also a range of other community-minded makerspaces in the EU. There is also a significant open-source movement in relation to 3D printing in the EU—with open-source companies such as Prusa 3D Printers.

In the face of the COVID-19 crisis, the 3D printing community in the EU sought to address some of the shortages in medical supplies. Commercial 3D printing companies particularly those with medical expertise—applied their skills and knowledge in additive manufacturing. A network of Fab Labs collaborated on community projects to help address some of the gaps in health equipment and medical supplies. Such interventions raised some larger questions in respect of intellectual property, innovation policy, and medical

4 of 32

regulation. The EU has been supportive more generally of a right to repair as a means of promoting sustainable development, a circular economy, and a Green New Deal.

2.1. The Italian Dispute

In spite of its modern health care system, Italy was one of the first nations overwhelmed by the COVID-19 crisis [45–47]. In this context, there were issues with shortages of medical suppliers—particularly in respect of medical supplies. In the midst of this public health crisis, volunteers engaged in the 3D printing of replacement valves for ventilators. There were allegations that the original manufacturers had threatened legal action for breach of intellectual property rights for making 3D printed versions of the replacement valves [48–51]. The murky facts of this controversy have been contested. While the media stories suggested that there had been threats of legal action for intellectual property infringement, the original manufacturer has denied making any such calls for litigation. The nature of the controversy has been lost in translation, as the Italian story has been converted into English stories in the popular press. What was left was a Rashomon-style story—with different protagonists providing various accounts of what happened.

Davide Sher provided one of the most thorough accounts of the 3D printing project, discussing the various efforts of the Maker Movement to provide help and assistance to the Italian hospital service [52]. Massimo Temporelli, founder of The FabLab in Milan, was contacted by Nunzia Vallini, editor of the Giornale di Brescia. She explained that the hospital in Brescia urgently needed valves for an intensive care device and that the supplier was unable to provide them in a short time. Temporelli contacted Cristian Fracassi—the founder and CEO of 3D printing company, Isinnova. Isinnova was able to use 3D printing to redesign and produce the missing piece. S.A. Applin highlighted the Italian effort as an example of Maker Culture ingenuity in dealing with the pandemic [4]. *Forbes* provided a profile of the Italian engineers who were 3D printing respirator parts for free [52].

Initially, there were media reports that there had been threats of intellectual property action against the 3D printing project [53]. This story was then amplified by international media [54–56]. Fracassi was reported as saying 'I have lawyers who are evaluating the matter, I am not dealing with it personally because I prefer to devote myself to this [the 3D designs]' [56]. Fracassi was also reported as saying 'In any case, I am holding my hands because in a world where money matters more than someone's health, nothing else can be done' [56]. There seemed to be some ambiguity in the media reports about the story of intellectual property infringement. There was uncertainty as to what regime of intellectual property was involved—it was unclear whether it was a patent matter or a copyright concern or a question of industrial designs. There was also a lack of clarity as to the value of the technology in question.

In an interview with *The Verge*, one of the 3D printing makers Alessandro Romaioli denied that there had been a threat of intellectual property infringement [50]. He observed that the company had refused to release design files, forcing them to reverse-engineer the valve. Romaioli commented: 'I talked to an operator who told me he couldn't give me the files, but after that we didn't receive anything from the original company—so I can assure you we didn't get any threat' [50]. Romaioli observed: 'They said they couldn't give us the file because it's company property, but that's all' [50]. The other 3D printing maker Cristian Fracassi told *Fast Company* over email that 'nobody is causing legal trouble or trying to sue us' for 3D printing these valves [57]. Fracassi also clarified that the original valves did not cost USD 10,000, despite what some outlets had been reporting [57].

The firm Intersurgical denied that the company had ever threatened legal action against the 3D printing of the valves. Managing director Charles Bellm made a statement to *The Verge*, insisting that 'recent reports from Italy are totally incorrect' [50]. Intersurgical have issued further denials of making any threats of intellectual property proceedings on media and social media [58]. The company stressed that 'recent reports from Italy are totally incorrect' [58]. Intersurgical maintained: 'We were contacted at the end of last week for manufacturing details of a valve accessory but could not supply these due to

medical manufacturing regulations' [58]. The company noted: 'We have categorically not threatened to sue anyone involved' [58]. Intersurgical observed that media reports had misrepresented the cost of the technology: 'The valve is an accessory supplied as part of a CPAP Hood system, which alone costs a few euros' [58]. Intersurgical noted: 'Our Italian company has been doing their utmost to supply the hospitals at this time and have been supplying these free of charge in many cases to use with the CPAP Hoods' [58]. The company lamented it had been the victim of false reporting, which had unfairly hurt the company's reputation: 'It's very disappointing that in the current climate this incorrect information is circulating' [58]. Intersurgical insisted: 'Our focus as a company is to be able to supply the hospitals that require these and many other vital products, and we are making every effort to ensure we can do so' [58].

Taking into account the protagonists' comments, fact-checkers have listed the media and social media stories that there was a threat of legal proceedings against the Italian 3D printing project as false [59].

Although the facts about the dispute seem garbled and confused, this controversy has nonetheless sparked a broader debate about intellectual property and the right to repair—particularly in the context of COVID-19.

2.2. 3D Printing and the Right to Repair

In his book, *The Right to Repair*, Professor Aaron Perzanowski comments that there are outstanding issues in respect of patent law and the right to repair ([17], p. 124–132). He specifically raised the risk of 3D-printing users facing patent infringement lawsuits: 'Nothing in patent law would prevent a more mercenary device maker from pursuing such a claim' ([17] p. 132).

Professor Lucas Osborn has focused on how intellectual property doctrine deals with new developments in 3D printing and additive manufacturing [22]. In an op-ed, Osborn commented that the controversy raised larger issues about intellectual property, 3D printing, and medical devices: 'Regardless, this episode represents the first widely publicized instance of 3D printing technology being used to (arguably) infringe a patent on a medical device' [60]. He was particularly interested in questions around patent infringement. Osborn noted that 'anyone who prints the physical valve commits direct infringement for "making" the patented device' [60]. He wondered whether there should be protection for intermediaries in the network for 3D printing. Osborn highlighted the reach of indirect patent infringement: 'A person can be liable for indirect patent infringement for helping or inducing others to commit direct patent infringement' [60]. Osborn commented that the 'creators of 3D printable files, especially those with knowledge of a relevant patent, should be wary of making them available for others on the internet' [60].

Osborn commented, though, that there were reputational costs to patent litigation: 'Even if the patent holder is more interested in making money than saving lives, it may be wise to consider the reputational and other costs associated with denying live saving equipment to hospitals in need' [60]. He noted that patent lawsuits in respect of COVID-19 vaccines had faced a backlash: 'Others, including patent holders relating to vaccine development, have initially threatened patent infringement suits only to backtrack after a storm of public outrage' [60].

Osborn observed that the controversy in Italy raised larger questions about the international framework for intellectual property and public health: 'Although the coronavirus pandemic inflames passions when needed medical equipment is in short supply, it is important to remember that in emergencies Article 31 of [the *TRIPS Agreement* 1994] [61], the key international patent treaty, provides flexibilities for governments to use—and authorize others to use—patents without the consent of patent holders' [60].

Considering the case study, Professor Jorge Contreras noted: 'While the existence of the threat and the patents remains murky, the incident sparked legal commentary regarding the risk that volunteers fabricating parts for lifesaving devices, and the hospitals that use

them, could be liable for patent infringement' [62]. He suggested that the matter raised larger questions of patent exceptions—and the question of the right to repair.

Reflecting upon the Italian controversy, Glyn Moody commented: 'Whatever the details, the episode underlines why the 3D files of these kind of devices should be made available routinely to hospitals' [63]. He observed: 'That would allow them print in cases of urgent need, regardless of any claimed patents, so that this kind of situation doesn't arise at all, and lives are not put at risk' [63]. Moody highlights the need for a better articulation of intellectual property exceptions to deal with public health emergencies in the European Union.

There has also been some discussion as to the operation of compulsory licensing under patent law in Italy [64].

Leading European scholar Professor Rosa Ballardini and her colleagues maintained that there should be further intellectual property flexibilities to deal with 3D printing and medical emergencies [65]. The authors concluded that 'it is our hope that the current focus on the role of IP will trigger more nuanced debates that will ultimately enable us to improve pandemic preparedness, as well as to develop more innovative and sustainable pandemic responses for many sectors, including 3D [Printing]' ([65] p. 1168). They observed that there was a need to make the most of the potential of 3D printing and distributed manufacturing: 'Approaches to tackling this and any future crisis may herald a new era of more sustainable digital distribution and trade' ([65] p. 1169).

2.3. Policy Issues

There has been a notable push in the European Union for stronger recognition of the right to repair—particularly to help promote a circular economy and sustainable development [66–68].

The right to repair movement maintained that there was a need for greater policy action on the right to repair during the COVID-19 crisis. Chloé Mikolajczak—a campaigner with Right to Repair Europe—said that the pandemic raised larger questions about the importance of repair [69]. She noted: 'During the pandemic, getting your device or appliance fixed is even more important, at a time when buying a new one might not even be possible, or we might simply not be able to afford it' [69]. She was concerned that 'in several European countries repair has not made the list of "essential activities" that should remain open in these times of crisis' [69]. Mikolajczak was concerned that independent repairers 'increasingly face legal threats' [69]. She argued that 'this criminalisation of independent repair businesses providing essential services needs to stop' [69].

Mikolajczak observed that there was a need for further policy reform post-pandemic: 'We must ensure that the independent repair and reuse sectors are not forgotten in the wider policy conversations about what the post COVID-19 world will look like and how to finance it' [69]. She stressed: 'Not only repair and reuse are essential to reach our climate and sustainability ambitions, both at national and EU levels; they also provide opportunities for jobs and training that will likely be much needed in the next few months and years' [69].

In addition to the Maker Movement, there were a number of European 3D printing companies who contributed towards the public health effort in respect of COVID-19. Materialise NV—based in Belgium—was involved in various initiatives: 'During the COVID-19 pandemic, we drew on the advantages of 3D printing to develop new solutions to support healthcare systems and help individuals stay safe' [70,71]. In particular, Materialise highlighted its contribution to the development of critical medical devices, personal protective equipment, and prevention devices. Materialise emphasized the benefits of distributed manufacturing for hospitals during public health emergencies—such as COVID-19.

The Dutch-American Company Shapeways also contributed to the COVID-19 relief effort [72]. Shapeways observed: 'Many members of the 3D printing community have set to work producing protective wear for medical personnel, as well as anti-contamination accessories and more' [72]. Shapeways noted that 3D printing was better suited to the production of certain technologies: 'Though not everything can be fully 3D printed due to FDA regulations and the complexity of medical equipment, 3D printing offers a fast prototyping and production process so that it may step in where traditional manufacturing falls short' [72]. Shapeways, in particular, highlighted the utility of 3D printing in the production of valves and ventilator parts; snorkeling mask ventilators; face shields; test swabs; face masks; door opening accessories; and quarantine booths.

The Czech open-source company Prusa 3D Printing sought to provide open-source designs for 3D printing personal protective equipment [73–75]. Prusa 3D printing emphasized that open licensing would enable 3D printing of personal protective equipment to take place—without any fear of intellectual property reprisals. Prusa 3D Printing also obtained approval for large-scale production of face shields with approval from the Czech Ministry of Health. Josef Prusa commented that the 3D printing community's response to the COVID-19 crisis highlighted the social value of the technology: 'This is a chance for our desktop 3-D community to show as a mature community, not just a bunch of nerds printing toys' [76].

At an industry level, additive manufacturing organizations discussed the importance of 3D printing during the COVID-19 crisis. CECIMO is an umbrella organization that serves the common interests and values of the European machine tool industries and related manufacturing technologies in the EU and at a global level [77]. CECIMO commented: 'The beginning of the COVID-19 crisis posed a challenge to the usual production and distribution channels for medical devices' [77]. CECIMO observed: 'Due to the scale of the demand, the conventional suppliers of this essential equipment were unable to provide immediate solutions to hospitals' [77]. CECIMO highlighted the importance role of additive manufacturing during the COVID-19 crisis: 'Amid this significant disruption in the supply chain, the additive manufacturing (AM) community stepped up and helped cover this surplus demand, providing access to equipment as well as other services' [77].

CECIMO made a number of recommendations to European policy-makers. First, the industry body emphasized that the COVID-19 crisis highlighted 'the need for dedicated AM standards and certification' [77]. The organization observed: 'It is essential to fast-track the development of standards and certification procedures to enable the use of AM in different areas and consequently allow the entire sector to expand' [77]. Second, the industry body noted that the COVID-19 crisis highlighted the 'Importance of rethinking conventional supply chain' [77]. Third, CECIMO commented that 'regulations should focus on unleashing technology's potential' [77]. The organization suggested that the 'EU industry strategy could boost the integration of AM in more sectors, ensuring a more resilient and competitive manufacturing industry in Europe' [77]. Finally, CECIMO noted that the 'lack of skilled workforce can slow-down AM growth' [77]. The industry body noted that 'the quick adoption of AM solutions in hospitals will result in an immediate demand of new specialized workforce' [77]. CECIMO commented that it was 'essential to expand the pool of European workers who are able to work with the full process from design to final parts in different sectors' [77].

3. The United States of America

In spite of being one of the most industrialized nations, the United States was vulnerable to the COVID-19 crisis. It has been reported from 2020 to the beginning of 2023 there were 101,496,168 confirmed cases of COVID-19 with 1,103,936 deaths reported to the WHO [78]. During the first onslaught of COVID-19, the US suffered from shortages in respect of PPE and other medical supplies. There was a significant debate about the ability of the 3D printing community to make and repair medical products in the US.

3.1. The Maker Movement

In March 2020, the US healthcare community experienced profound shortages of PPE and associated accessories, and medical devices to treat COVID-19 patients [79]. There was also a pressing need to fix and repair broken medical equipment, such as ventilators [80].

Founder and editor of *Make Magazine* Dale Dougherty (with Victor Hwang) discussed how 'a grassroots uprising of makers, engineers, and others' are creating a backup plan for the COVID-19 response [81]. They observed: 'In the face of COVID-19, shortages of medical equipment, such as ventilators for patients, and protective gear for personnel in hospitals are becoming a critical problem' [81]. Dougherty and Hwang noted that 'Plan A' involved governments using their powers to produce needed equipment, while Plan B involved private industry stepping up to produce equipment and supplies in their factories [81]. Dougherty and Hwang suggested: 'The Maker Movement might provide a "Plan C" for America and the world' [81]. The authors commented: 'This uprising of action in response to COVID-19 demonstrates the ingenuity and talent that flourishes at the grassroots, alongside the resources of government and the corporate sector' [81]. Dougherty and Hwang hoped that such grassroots innovation by the Maker Movement may be useful for addressing other global development challenges, such as those presented by the UN SDGs.

Dale Dougherty and Representative Tim Ryan (a Democrat from Ohio who was part of the Maker Caucus) called for the Maker Movement to play a key role in the COVID-19 as well [82]. Dougherty and Ryan invoked the historical example of the New Deal program of the Civilian Conservation Corps (CCC) during the Great Depression. Dougherty and Ryan argued: 'It is time for a new version of the CCC, one that coordinates grassroots efforts, provides more training for others to participate, and creates a civic infrastructure that can make our country more resilient in the future' [82]. Dougherty and Ryan suggested: 'Today, we propose launching the Civic Response Corps (CRC), a new program in the spirit of the CCC to coordinate local civic response efforts, train the unemployed, undereducated, and unskilled to participate and create a new civic infrastructure needed not only to respond to the crisis but ramp up the recovery' [82]. They highlighted that 'citizen-makers are rising to the challenge to create the medical supplies needed to address shortages in our local communities' [82]. (Ryan has since left the US Congress in 2023, after his House seat was redistricted, and he made an unsuccessful effort to win a US Senate seat in Ohio).

In addition to community efforts in respect of 3D Printing PPE, industry played an important role with industrial 3D printing. A leading commercial 3D printing company, HP, has been producing critical parts to meet urgent needs—including face shields, masks, personal accessories, and ventilator components [83]. Electronics manufacturing company and open-source hardware design company Adafruit Industries became involved in crisis relief in 2020 after New York City officials put a call out for the local manufacturing businesses to address shortages in personal protective equipment and medical devices [84,85]. A pioneer of the 3D printing industry, Avi Reichental, observed that the crisis had shown the power of open innovation and crowdsourced design: 'One lesson that has been undeniably learned is rooted in the immense power of crowd-sourced design: from homemade cloth masks distributed via online crafting sites to innovative solutions for social distancing, some of the most effective, cost-effective and elegant quick fixes to the staggering challenges created by this pandemic came as a result of decentralized, designers with high creativity and a desire to jump in and help' [86]. Nonetheless, Reichental noted that 'there have also been some hard lessons learned about the need for regulation and oversight, even in an industry like ours which is powered so strongly by the free exchange of ideas' [86].

In the educational sector, there was also some useful applications of 3D printing during the COVID-19 crisis. The University of Miami used 3D printing to develop a reusable N95 mask [87]. Helpfully, New Jersey Institute of Technology (NJIT) joined the Open COVID Pledge [88], making its test swab technology widely available free of licensing fees [89]. Patent pledges and open-source licensing could alleviate some of the concerns about the risks of intellectual property infringement [90].

Professor Joshua Pearce—who has run labs at Michigan Tech and Western University has advocated a model of open medicine [91]. He has published a roadmap for how opensource medical supplies could be used to address public health pandemics, as well as a possible roadmap for establishing open-source protocols for such lifesaving technology [91]. Pearce has contended: 'Distributed digital manufacturing offers a solution to medical supply and technology shortages during pandemics' [91].

The US Patent and Trademark Office issued an award under its Patents for Humanity scheme to the University of South Florida Health, Northwell Health, Tampa General Hospital, and Formlabs for 3D printed nasal swabs for use during the COVID-19 pandemic [92,93]. There has been a debate, though, whether the Patents for Humanity scheme is only a weak incentive for encouraging humanitarian innovation [94].

The US Food and Drug Administration worked with the National Institutes of Health and Veteran Affairs Department to work together to facilitate regulatory and basic science innovation with 3D printing technologies to respond to COVID-19 [95,96]. Dale Dougherty and Tim Ryan observed that America Makes was organizing community efforts to engage in 3D printing of PPE and medical supplies: 'In Youngstown, Ohio, the national institute for additive manufacturing, America Makes, is coordinating these community-driven efforts to create new, easily-manufactured designs for masks, face shields, ventilator parts, and dozens of other open-source ideas to help protect our front-line medical workers, prevent the spread of infection, and save lives' [82]. America Makes played a critical, key role in coordinating public and private efforts to deploy additive manufacturing in support of the U.S. COVID-19 response [79]. The COVID 3D Trusted Repository for Users and Suppliers through Testing (COVID 3D TRUST) acted as an initiative to gather and test open-source designs for 3D printing PPE and devices [97].

Discussing the situation in the US, Matthew Bultman observed that 'innovators and volunteers are rallying to 3D printing to combat the new coronavirus' but noted that 'with this ingenuity comes concerns about patent infringement and product safety' [98]. He highlighted: 'Owners of patents on certain designs of face shields, masks, and ventilator parts could have infringement claims against printers' [98]. Bultman also noted: 'There is also a risk of lawsuits if supplies are unsafe' [98]. Professor Lucas Osborn commented: 'If there's a patent covering it and you actually 3D print it, then you're clearly infringing' [98]. Attorney Graham Phero commented that 'it's fantastic to have everybody come together' but 'we really do need to respect patent rights and we do need to make sure these products are safe for the end consumers' [98].

3.2. The Medical Right to Repair

In 2020, there was discussion in the US about the need to address the right to repair in light of the public health emergency in respect of COVID-19.

A number of civil society groups—including US PIRG, Repair.Org, and iFixit—supported a petition for a right to repair. The petition read as follows: 'U.S. hospitals do not have enough ventilators to meet the spike in cases of respiratory failure that the novel coronavirus is projected to create' [99]. The petition emphasized that 'it will become critical to remove barriers to repairing ventilators' and 'we may need to repair older reserve ventilators so they can be put into service' [99]. As the petition demanded, 'I urge you to immediately release service information—manuals, access to error logs and diagnostic information or other repair resources—for hospital ventilators to help our hospitals combat the coronavirus' [99].

Nathan Proctor, Right to Repair campaign director with U.S. PIRG, said that 'Right now, ventilator repair and maintenance issues are life and death issues' [99]. He contended: 'Manufacturers of ventilators should immediately release service manuals, service keys, schematics and service keys' [99]. Proctor stressed: 'Lives are at stake—This is no time to be proprietary' [99]. Proctor commented: 'From removing barriers to repair to ramping up production, there are steps we must take now to maximize ventilator supply and save lives' [99]. Kevin O'Reilly of PIRG added: 'There is no reason we should tolerate manufacturers putting their own proprietary concerns over patient safety—especially during the pandemic' [100]. He called for Congressional recognition of the right to repair: 'Passing this bill is an easy, common-sense way for the Senate to help hospitals in their time of need, and a terrific first step towards a permanent solution' [100]. Gay Gordon-Byrne of Repair.org commented: 'We've been fighting for our right to repair medical technology since our founding seven years ago' [99]. The executive director argued: 'The coronavirus pandemic has proven that manufacturer restrictions on access to repair information and technicians is a recipe for disaster' [99]. Kyle Wiens, iFixit.com CEO, commented: 'A single hospital might have ventilators made by four different manufacturers and it can be a headache trying to find the right information, so iFixit is trying to help make that easier' [99]. He observed: 'We want to make sure that a technician doesn't have to hunt for these manuals—every second counts right now' [99].

Kathleen Burke of Public Knowledge was concerned about the relationship between intellectual property and the right to repair during the COVID-19 crisis: 'If even hospitals and states during a state-of-emergency face outcry over side-stepping IP barriers to repairing goods that they own, then we have obviously created a system that is broken' [101]. Burke maintained: 'As we rethink how our world is shaped in light of COVID-19, making sure consumers have a meaningful right to repair is one issue space that can help ensure our world is more sustainable' [101]. John Bergmayer, Legal Director at Public Knowledge, supported the call for legislative action: 'Especially during the pandemic, it's important that medical services aren't interrupted by expensive, slow, and unnecessary service requirements' [102].

There have been concerns that intellectual property owners have been hampering repairs in respect of critical medical infrastructure and equipment during the COVID-19 pandemic [103].

Democrats Oregon Senator Ron Wyden and New York Representative Yvette D. Clarke have introduced right to repair legislation in the Senate and the House of Representatives that would make it easier for hospitals to fix medical equipment during the COVID-19 pandemic. Oregon Senator Ron Wyden maintained: 'There is no excuse for leaving hospitals and patients stranded without necessary equipment during the most widespread pandemic to hit the U.S. in 100 years' [100]. He commented: 'It is just common sense to say that qualified technicians should be allowed to make emergency repairs or do preventative maintenance, and not have their hands tied by overly restrictive contracts and copyright laws, until this crisis is over' [100]. New York Representative Yvette D. Clarke observed: 'As America grapples with this lethal pandemic, we are also experiencing unprecedented shortages of medical equipment' [100]. She contended that it was critical to expand access to life-saving devices: 'This narrowly-tailored, common-sense, and timelimited bill will ensure critical medical items like ventilators do not go to waste due to maintenance restrictions that have no nexus to safety' [100].

The United States Congress has been considering the *Critical Medical Infrastructure Right-to-Repair Act* of 2020 (US) [104,105].

The legislation is a bill 'to amend title 17, United States Code, to address circumvention of copyright protection systems with respect to the maintenance or repair of critical medical infrastructure, and for other purposes'. Section 1 provides that the short title of the Act is the *Critical Medical Infrastructure Right-to-Repair Act* of 2020. Section 2 provides definitions in respect of the bill—including in respect to the terms 'commerce', 'covered emergency', 'covered service provider', 'critical medical infrastructure', 'repair', and 'service material', as well as 'critical media infrastructure contract', 'service provider', and 'trade secret'.

Section 3 provides for a new clause in the *Copyright Act* 1976 (US)—section 123, which would provide for a 'Limitation on exclusive rights: incidental copies of service materials made during maintenance or repair of critical medical infrastructure'. Section 3 also provides for an amendment to section 1201 of the *Copyright Law* to allow for the permissible circumvention of a technological protection measure 'to repair or maintain critical medical infrastructure with respect to that covered healthcare provider' as 'part of preparation for, or a response to, the covered emergency'.

This safeguard seems to be a response to copyright litigation over repair manuals during the COVID-19 crisis. In the United States, there have also been copyright threats over repair manuals during the coronavirus public health pandemic [106]. Steris sent a

letter to iFixit asking the repair organization to remove from its website repair information for Steris equipment. iFixit had published the medical device repair information in order to help hospitals and other medical organizations through the COVID-19 pandemic. In response, the Electronic Frontier Foundation responded to Steris on behalf of its client, iFixit [107]. First, the legal advocacy group noted that 'iFixit is protected by Section 512 of the *Digital Millennium Copyright Act*, which allows online platforms to host content contributed by users provided they comply with the Act's requirements, which iFixit does'. Second, the EFF maintained that iFixit was protected under the defense of fair use: 'The fair use doctrine authorizes iFixit and contributors to the Database to share the repair information they are providing' [107]. The EFF considered the various factors involved in a fair use determination, and maintained that such factors favored iFixit. The EFF concluded: 'The Medical Device Repair Database promotes the public interest by improving access to information to help technicians, and the strapped hospitals they work for, do their jobs more effectively' [107]. The EFF commented: 'Given that the market for medical devices is about medical devices, it would be difficult for Steris to plausibly argue that it lacks adequate other incentives to document how to maintain the devices that are its bread and butter' [107]. The EFF concluded: 'The benefit to the public far outweighs any speculative harm to any legitimate interest in restricting their availability via the Database' [107].

It should be noted that there has also been significant copyright litigation against the Internet Archive for making electronic books available in a National Emergency Library during the COVID-19 lockdown [108].

Section 4 of the legislation amends Section 271 of the design patents regime. In the new legislative provision, 'It shall not be an act of infringement with respect to a patent for design obtained under section 171 for a covered healthcare provider to fabricate a part on a non-commercial basis, and as needed, for the repair or maintenance of critical medical infrastructure with respect to that covered healthcare provider, if the repair or maintenance is part of a response to the covered emergency'.

Section 5 deals with contracts. The new provision seeks to prevent the contracting out of the right to repair: 'Notwithstanding any other provision of law or regulation, a provision of a critical medical infrastructure contract is null and void if that provision of the critical medical infrastructure contract prohibits or restricts the ability of a covered healthcare provider that is a party to the contract to, in response to the covered emergency, repair or maintain critical medical infrastructure with respect to the covered healthcare provider'.

Section 6 focuses on manufacturer requirements.

Section 7 asks the Federal Trade Commission, in consultation with the Register of Copyrights and the Under Secretary of Commerce for Intellectual Property and Director of the United States Patent and Trademark Office, to conduct a study with respect to innovation and anticompetitive practices in the market for critical medical infrastructure.

The House bill has been referred to the House of Representatives Committee on the Judiciary in addition to the Committee on Energy and Commerce. The Senate bill has been referred to the Senate Committee on the Judiciary.

Oregon Senator Ron Wyden wrote an op-ed with the President of the American College of Clinical Engineering, Ilir Kullolli, on the need for the United States Congress to support the bill [109]. They highlighted: 'As the crisis continues, concerns about the maintenance of critical medical equipment, including X-ray machines, dialysis machines, and ventilators, are growing' [109]. Wyden and Kullolli discuss some of the barriers and obstacles to repair of medical equipment during the COVID-19 public health emergency. They despaired: 'Too many hospitals face long waits for authorized technicians to repair life-saving machines' [109]. They lamented: 'A recent study shows that an overwhelming majority of repair technicians have been blocked from making critical repairs as a result of manufacturer restrictions' [109]. Wyden and Kullolli commented: 'This commonsense legislation would allow trained repair technicians to more easily access the information and tools they need to fix and maintain critical medical infrastructure during the

COVID-19 crisis' [109]. Wyden and Kulloli denied that the legislative regime would adversely affect safety standards. Wyden and Kullolli also supported a more general right of repair: 'While this bill is focused on addressing the pandemic-induced emergency facing medical infrastructure, Americans should always have the right to repair the vehicles, tools, and devices they own' [109]. Wyden and Kullolli commented: 'Our hope that this legislation serves as the first step toward establishing a new balance that strikes down unnecessary obstacles to home repair—everything from tractors to electronics—while continuing to allow manufacturers to innovate and thrive' [109].

Various civil society and public interest advocates supported the bill [100]. Alan Morgan, CEO of the National Rural Health Association said: 'As COVID-19 surges across rural America, rural providers must have the rapid ability to maintain effective and operational equipment' [100]. Color Of Change Vice President Arisha Michelle Hatch commented: 'Since the onset of the COVID-19 pandemic in the United States, Color Of Change has pushed ventilator manufacturers to dial-back their dangerous, counterproductive repair restrictions, which have put an unnecessary strain on our medical providers' ability to tackle the virus' [100]. The Critical Medical Infrastructure Right-to-Repair Act of 2020 (US) has been supported by a spectrum of health care, engineering, and civil society groups. The legislation received endorsements from the American College of Clinical Engineering (ACCE); Association of Medical Service Providers (AMSP); National Rural Health Association (NRHA); National Association of Rural Health Clinics (NARHC); International Association of Medical Equipment Remarketers and Servicers (IAMERS); Alliance for Quality Medical Device Servicing (AQMDS); ISS Solutions Healthcare Technology Management; U.S. Public Interest Research Group (U.S. PIRG); The Repair Association; the Electronic Frontier Foundation (EFF); Color of Change; Public Knowledge; R Street Institute; Re:Create; Lincoln Network; Niskanen Center; Colorado Association of Biomedical Equipment Technicians (CABET); MaineGeneral Medical Center; Pennsylvania Public Interest Research Group (PennPIRG); and Center for Democracy & Technology [110]. R Street supported the bill [111]. According to R Street Distinguished Senior Fellow Mike Godwin, 'This carefully crafted bill allows technical personnel to lawfully repair medical infrastructure and equipment during the COVID-19 pandemic, while also ensuring patent holders control commercial use of their intellectual property per their statutorily defined rights' [111]. The Niskanen Center supported the proposed legislation [112]. Daniel Takash, regulatory policy fellow at the Niskanen Center, was concerned that copyright law and technological protection measures were putting in place unnecessary barriers: 'The consequences of these policies go from excessive to dire if we're talking about medical equipment like ventilators' [112].

Benjamin Louviere has argued: 'The COVID-19 pandemic has cast into sharp relief the already pressing need for Congress to intervene and pass right-to-repair legislation, at the very least with respect to medical devices for hospitals' [113].

In addition to law reform, Professor Jorge Contreras of the University of Utah has called on courts to take a liberal view of the legitimacy of repair in any patent disputes during the pandemic: 'In order to permit needed repairs and parts replacements for critical health-related equipment, courts should take a liberal view of the repair doctrine' [114]. He contended that 'the immunity from suit afforded by the repair right should be extended not only to the owners of patented equipment, but to their suppliers, parts vendors and maintenance organizations' [114]. Contreras has maintained that there is scope for expanding exceptions to patent infringement for research and repair in response to the pandemic [62].

Scholars Ofer Tur-Sinai and Leah Chan Grinvald have contended that 'the need for a right to repair medical equipment is particularly evident during this challenging period' [115]. They added: 'Carving out a safe legal space for repair would serve important policy considerations during "normal" times and increase the preparedness of our legal system' [115].

Shuhan He (of GetUsPPE), Debbie Lai (of CovidActNow), and Jarone Lee endorsed the proposal for a medical right to repair in *The Lancet* [116]. The correspondents supported

13 of 32

Senator Ron Wyden's bill: 'During these extraordinary times, such legislation for the right to repair not only moves the medical field in a more affordable, efficient, and sustainable direction but also enables life-saving services to continue to be available at times of high stress' [116].

The *Critical Medical Infrastructure Right-to-Repair Act* of 2020 (US) H.R. 7956—and its companions—have not progressed through the United States Congress. In the intervening time, there has been organized resistance from medical device makers to the proposal for a medical right to repair. The Advanced Medical Technology Association (AdvaMed) has been a notable opponent of a medical right to repair, arguing that it would undermine health and safety [117]. As well as opposing the Federal bill, AdvaMed has argued that state right to repair laws should not extend to medical technologies and devices [118]. The Medical Imaging & Technology Alliance (MITA) has also been a critic of proposals to extend the right to repair to medical devices [119]. MITA has maintained that a medical right to repair would adversely affect the cost, functionality, and lifetime upkeep of medical technologies and devices [120]. Henry Miller of the Center for Medical Economics and Innovation argued that a medical right to repair would undermine intellectual property and cybersecurity [121]. There are also, more broadly, a variety of technology developers and industry and trade associations who are adamantly opposed to legislative proposals on the right to repair (whatever form they may take).

3.3. Other Right to Repair Proposals

The US policy debate over the right to repair is a longstanding one [122]. The current policy debate over the right to repair has been a dynamic and volatile one. While the Wyden bill for the medical right to repair has languished in the United States Congress, there are a range of other state and Federal proposals on the general right to repair, which have advanced further.

In 2021, Rep. Joseph Morelle put forward a private member's bill, the *Fair Repair Act*, to provide recognition of a right to repair [123]. This legislation would require an original equipment manufacturer (OEM) to make diagnostic, maintenance, and repair equipment available to independent repair providers. Morelle has emphasized that 'For too long, large corporations have hindered the progress of small business owners and everyday Americans by preventing them from the right to repair their own equipment' [124]. As he contended, 'This common-sense legislation will help make technology repairs more accessible and affordable for items from cell phones to laptops to farm equipment, finally giving individuals the autonomy they deserve' [124].

In 2022, House Representatives Mondaire Jones, a Democrat, and Republican Victoria Spartz introduced the *Freedom to Repair Act* to reform copyright law to make it easier for consumers to obtain repairs [125]. Under this legislation, the copyright regime would not prohibit an act of circumvention or trafficking for the purpose of diagnosing, maintaining, or repairing electronic equipment. However, this legislative proposal explicitly excludes the manufacturer or distributor of a medical device.

There is also an *Agricultural Right to Repair Act* sponsored by Senator Jon Tester [126]. There is a bipartisan bill put forward by Republican Congressman Darrell Issa and various Democrat representatives to reduce car repair costs [127]. There are several other bills that have been prepared at a Federal level. There have been proposals for other specialist reforms, such as a right to repair software [128].

In July 2023, the House Judiciary Subcommittee on Courts, Intellectual Property, and the Internet held a hearing on the topic 'Is There a Right to Repair?' The Committee led by Chairman Issa considered a range of stakeholder views on the right to repair [129]. Professor Aaron Perzanowski made the case for a broad recognition of a right to repair, whilst Devlin Hartline of the Hudson Institute argued that the right to repair was a legal fiction. There was also evidence from Scott Benavidez from the Automotive Service Association, Paul Roberts of SecuRepairs, and Kyle Wiens of iFixit. While there was a wide ranging discussion

of the topic of the right to repair, the issue of a medical right to repair was only mentioned in passing.

For its part, the Biden Administration has shown enthusiasm for the recognition of a general right to repair. President Joe Biden has issued an executive order, highlighting that the right to repair is a priority across government [130]. The Federal Trade Commission under Lina Khan has taken a number of enforcement actions in respect of the right to repair (including in the field of health-care) [131]. Moreover, the Federal Trade Commission under Lina Khan has promised further law reform and policy development on the topic of the right to repair [132].

There have been a range of proposals for a right to repair at a state level in the United States. In addition to the Federal *Critical Medical Infrastructure Right-to-Repair Act* of 2020 (US) H.R. 7956, there has also been a specific bill in California focused on a medical right to repair put forward by Democrat Senator Susan Eggman—the *Medical Device Right to Repair Act* (SB 605) [133]. This Californian legislative proposal would require an original manufacturer of powered medical equipment used in the treatment, monitoring, or diagnosis of a patient to provide documentation, parts, service access methods, and tools used to inspect, diagnose, maintain, and repair powered medical equipment to a hospital and an independent service organization engaged by the hospital for the purpose of providing medical equipment maintenance and repair, on fair and reasonable terms, as defined. The bill would subject an original equipment manufacturer who knowingly violates these provisions to specified civil penalties. The bill would exempt any trade secret information from these requirements. The Appropriations Committee failed to pass the California Medical Right to Repair Act [134].

4. Australia

In some respects, Australia had a strong public health response to the COVID-19 crisis, at least initially. According to the WHO, Australia had between 2020 and early 2023 a total of 11,326,032 cases of COVID-19, with 18,190 deaths [135]. Nonetheless, Australia certainly experienced issues in obtaining timely access to essential medicines. Moreover, Australia also had severe shortages of PPE and other medical equipment at the outset of the COVID-19 crisis. During the COVID-19 crisis, the Productivity Commission investigated the question of the right to repair. The Australian Government has begun to implement some of the Productivity Commission's recommendations on the right to repair. However, the topic of the medical right to repair is still in a state of abeyance.

4.1. 3D Printing Response

In 2020, at the beginning of the COVID-19 crisis, the Australian Government sought to map domestic production capability of medical personal protective equipment (PPE) [136]. In particular, the Australian Government sought 'information on domestic production capabilities relevant to a range of medical PPE, including surgical gowns, gloves, goggles, hand sanitisers, clinical waste bags, waste bag closure devices (ties), blood and fluid spill kits, mask fit test kits and thermometers' [136]. In contrast to the focal point of the development of the Canadian Shield in Canada [137], the Australian efforts to provide PPE were somewhat more dispersed and distributed.

In this context, there was activity within the 3D printing community in respect of addressing some of the shortages in respect of PPE. Australian universities and educational institutions were particularly prominent in terms of providing additional capacity for the production of PPE.

The University of Melbourne worked closely with Melbourne hospitals on 3D printing face shields [138]. The Maker Spaces team from the Melbourne School of Design tested designs for face shields. The Melbourne School of Engineering 3D Innovation Centre provided the capacity to print large quantities of 3D printed face masks. Researchers and engineers at the University of Melbourne also worked to develop isolation hoods,

low-cost ventilators, reusable N95 masks, and COVID-19 testing swabs (with the help of 3D printing).

Professor David Grayden commented: 'It's clear that these innovations are something hospitals really need' [138]. He observed that 3D printing enabled open-source collaboration and sharing: 'What COVID-19 is really drawing to the fore is that 3D printing can be used to prototype and rapidly manufacture small batches. If we need it tomorrow, 3D printing is a key way to do that. COVID-19 is showing us that we need to speed up the process between idea and manufacturing' [138].

Dr. Jasamine Coles-Black commented: 'Our vision is to have everyone in the Australian healthcare space linked up, and for no-one to go without PPE' [138]. She was conscious about larger questions in respect of safety, regulation, and intellectual property: 'We don't want to be in a situation where regulatory and administrative roadblocks prevent a validated solution from being effectively delivered to the front lines within an acceptable timeframe' [138]. Coles-Black highlighted larger ethical considerations about access to health care: 'We took an oath to look after our patients and our colleagues' [138].

The University of Wollongong has a strong reputation for its work in bioprinting and medical 3D printing. The University of Wollongong pivoted some of its 3D printing operations to 3D print medical supplies to support healthcare workers during the COVID-19 pandemic [139]. The University of Wollongong Makerspace manager Jessica Grozdanov said the face shield initiative is supported by international open-source collaboration: 'Over the past week, an open source design for a face shield that can be 3D printed has been circulated internationally' [139]. She commented: 'We have 3D printers available at the UOW Makerspace which got us thinking whether our local hospitals could benefit from an initiative like this' [139]. Grozdanov commented: 'Makerspaces and other innovative manufacturing initiatives are increasingly playing an important role in the response to crises, as we can respond quickly to a particular need and can offer innovative, local solutions in high priority scenarios' [139]. In addition, the various centers and facilities focused on bioprinting at the University of Wollongong also provided help and assistance. Professor Gordon Wallace noted: 'We're working towards a common goal and can move quickly, which is critical as face shields are in high demand, there is a great shortage of them and they are needed immediately' [139]. He observed: 'We're here and ready to help' [140].

Researchers at the University of New South Wales also worked on prototyping PPE designs in preparation for the COVID-19 pandemic. As Dr. Blake Cochran explained, 'We have been able to provide some of our early prototypes to our clinicians and to our medical researchers that are working on the virus and we are incorporating their feedback into improving our design' [141]. He reflected: 'So [we are] focusing on things such as making sure that our masks are comfortable, that they can be worn for long periods of time, they don't fog up, and that they are actually suitable moving forward' [141].

The University of Technology Sydney sought to engage in 3D printing of face shields for the benefit of Australia's neighbor, Papua New Guinea [142]. The UTS ProtoSpace encouraged COVID-19 student responders to help produce 300 health shields for medical personnel in Papua New Guinea. ProtoSpace intern Isaac Garcia commented: 'We had a number of channels for people to explore different types of personal protective equipment (PPE) before settling on face shields, as this is an example of the capability of 3D printing and of our capacity to achieve a quick response' [142]. ProtoSpace used an open-sourced design, which was easy to assemble and met Australian health regulations for PPE.

Australian Doctors International Program Officer Mark Newcombe was grateful for supplies: 'We have been able to locate hand sanitisers, face masks, gowns, goggles, gloves—but could not find face shields anywhere, so when Stuart contacted us to consider a donation we were delighted' [142]. He commented: 'We were really impressed with the quality, well designed and well made' [142]. Newcombe noted: 'The face shield obviously delivers protection for the user, but it also extends the life span of face masks, which is important for any equipment in remote and hard to reach areas' [142].

The Australian National University was also involved in 3D printing protective equipment for health workers [143–145]. Dr. John Debs of the ANU MakerSpace commented: 'We see this as a community good from the University to supplement the usual supply chain of hospitals, general practices and other frontline health services' [144]. He observed: 'We had heard buzz around people 3D printing face shields but, after looking at the designs, we knew we could make them faster and more effectively with other tools' [144]. Debs reflected: 'We found some open-source designs from the United States, and we ran with them as a starting point' [144].

The Herston Biofabrication Institute and the public research organization CSIRO were involved in the production of PPE for the assistance of the Queensland state government [146]. Other partners in this endeavor included the Queensland University of Technology (QUT), the University of Queensland, Healthia Group, COVID-SOS, Konica Minolta, and Shapelabs [146].

In the social enterprise field, 3D printing organizations were able to extend their work into new health-related fields. Free 3D Hands is a charity based in Australia, which designs and manufactures 3D printed assistive devices for those with disabilities [147]. The lead engineer Mat Bowtell has relied upon open licensing to make such technologies available and accessible [148]. During the COVID-19 crisis, Free 3D Hands was able to manufacture thousands of face shields for healthcare workers, who were having problems sourcing personal protective equipment during widespread shortages [2]. Mat Bowtell told *The Guardian*: 'With 3D printing, we've been able to go from making hands to making face shields in a matter of, well, days' [2]. He observed: 'To completely revamp our line, that's how agile this technology is and how flexible it is' [2].

In addition, 3D printing companies provided assistance during the COVID-19 crisis. The Sydney-based 3D Printing Studios engaged in the local manufacture of nasal and throat swabs which were used in COVID-19 testing kits [149]. The Melbourne company Seen Technology and Sydney company Composite Images joined a network of 3D printing distributors who were printing PPE [150]. The Australian 3D Manufacturing Association provided support for the community-based efforts of maker societies to engage in the 3D printing of PPE [151].

The 3D printing lab BioFab3D deployed its resources to help make facial shields [141]. Engineer and lab manager Cathal O'Connell commented: 'We are producing about 20 units a day of our face shield design which we supply to some of the clinics at St Vincent's Hospital [Melbourne]' [141]. She commented: 'Our main role being a fabrication lab based within a hospital is that we can try multiple designs with the clinicians themselves, get their approval, and then send out the approved design to the big 3D printing sites who can manufacture them in numbers of hundreds' [141].

There was a larger discussion about how Australia's situation highlighted the risks of offshoring the production of PPE [2].

4.2. The Productivity Commission into the Right to Repair

There has been a longstanding debate in Australia over the right to repair—but it has often been focused on particular sectors, such as automobiles, and agricultural machinery. The COVID-19 crisis raised questions about the right to repair in the context of health-care. The ACT Attorney-General Shane Rattenbury pushed for an inquiry into a right to repair in order to promote consumer rights, competition policy, sustainability, and a circular economy [152]. In the midst of the COVID-19 crisis, the Productivity Commission was asked to investigate the right to repair. The Productivity Commission produced an issues paper [153], a discussion paper [154], and a final report [155].

There were a range of submissions which considered the question of a medical right to repair.

The author of this article made a submission to the Productivity Commission on the right to repair—highlighting amongst other things, the need for a right to repair essential medical equipment, particularly during the COVID-19 crisis [156]. In this submission, it was

noted: 'The draft report by the Productivity Commission briefly discusses in passing some of the impacts of the coronavirus pandemic upon the topic of the right to repair' [156]. As this submission highlighted, 'It would be helpful and useful if the Productivity Commission could devote a chapter or a sub-chapter to the topic of public health and the right to repair (much like it has in respect of intellectual property, consumer rights, competition policy, product design, and e-waste)' [156]. The submission added: 'There has been much discussion of the necessity of law reform during the coronavirus emergency—including in respect of intellectual property and the right to repair' [156].

The author's colleague Dr. Muhammad Zaheer Abbas also made a submission to the Productivity Commission, observing that repair with the help of 3D printing technologies was critically important for medical technologies during the COVID-19 crisis [157]. He later expanded upon the need for patent exceptions for the right to repair to better enable 3D applications in response to the COVID-19 crisis [158].

Medical technology companies, though, have demanded to be exempted from any right to repair regime. For instance, Medtronic Australia argued: 'To ensure the continued safety and effectiveness of health restoring and life-saving medical devices, Medtronic believes medical devices should be excluded from consideration from "Right to Repair" [159]. As Medtronic Australia maintained, 'Medical devices designed, manufactured, and serviced by the manufacturer are categorically different than consumer and other goods in that they are regulated by the Therapeutic Goods Administration (TGA) throughout their lifecycle to ensure safe and effective operation for an intended use' [159]. Medtronic Australia warned that a "Right to Repair" could present dangerous unintended consequences for the patient with the key concern that it may compromise patient safety' [159]. Moreover, Medtronic Australia cautioned that a right to repair could compromise cybersecurity [159]. In conclusion, Medtronic Australia recommended that 'the Productivity Commission exclude medical devices and medical technology from the consideration of the "Right to Repair" [159]. (It should be noted, though, that Medtronic in the United States was willing to pledge intellectual property during the COVID-19 crisis) [90].

In its draft report, the Productivity Commission commented that 'although restrictions on repairs of medical equipment may generate some harm (particularly for any vulnerable or disadvantaged equipment users), this may not be sufficient to justify any policy response, due to elevated safety risks for some types of repair' [154].

In its final report, the Productivity Commission made a number of recommendations about the ways and means of realizing a right to repair in Australia [155]. The Productivity Commission made law reform recommendations in respect of consumer law and competition policy to support the right to repair. The Productivity Commission also made policy reform suggestions in respect of copyright law and the right to repair (but declined to make recommendations about other fields of intellectual property). The Productivity Commission also discussed questions around the right to repair, environmental stewardship, and sustainable development in Australia.

However, the Productivity Commission said that the topic of medical repairs required further investigation ([155] p. 144–146). This is an important and significant qualification to its overall law reform recommendations. The Productivity Commission acknowledged that the issue was a significant one: 'Several inquiry participants said that medical device manufacturers are unnecessarily restricting access to repair information and spare parts' ([155] p. 144). The Productivity Commission was conscious of the economic and health impact of restrictions on medical device repair: 'These repair restrictions can have detrimental impacts for both patients receiving health care (such as by delaying hospital procedures while equipment is awaiting repair) and for device users with a high dependency on their equipment (such as people who use a wheelchair or hearing aids)' ([155] p. 144).

Nonetheless, the Productivity Commission was conscious that there were special considerations in relation to the regulation of medical devices: 'Due to elevated safety risks, the medical device industry is also closely regulated by the *Therapeutic Goods Administration* (TGA), with many medical devices required to demonstrate that they conform with the

"essential principles" in the *Therapeutic Goods (Medical Devices) Regulations* 2002 before they can be supplied in Australia' ([155] p. 145). The Productivity Commission observed that 'it is not apparent that medical device regulations have found the right balance between repair access and device safety, and may be creating strong incentives for manufacturers to restrict repairs, generating harm to patients and device users through medical delays and additional costs' ([155] p. 145). The Productivity Commission held that 'due to the complexity of the medical device market and the wide variety of different devices covered by existing regulations, the Commission did not have sufficient evidence to justify specific policy changes' ([155] p. 145).

In Finding 4.5, the Productivity Commission found that 'medical device regulations do not consider repair access' ([155] p. 145). The Productivity Commission observed: 'Current regulations of medical devices—such as the "essential principles" in the *Therapeutic Goods* (*Medical Devices*) *Regulations* 2002—aim to minimise safety risks to patients and device users, which has the effect of encouraging manufacturers to restrict access to repair.' The Productivity Commission noted in its finding: 'The regulations do not appear to account for the potential harm from reduced access to repair services (such as medical delays and additional costs), or that risks are likely to be low for some devices or for repairs completed by highly-qualified independent repairers' ([155] p. 146). The Productivity Commission was conscious that there could be monopolies in respect of certain specialist medical devices: 'Time-sensitive services and user dependency also mean patients and device users often have limited alternative options, increasing their lock-in' ([155] p. 146).

In Recommendation 4.2, the Productivity Commission called upon the Federal Government to 'review the medical device market and regulations' ([155] p. 146). The Productivity Commission explained: 'The Australian Government should conduct an independent public review of existing medical device regulations to assess whether they strike a balance between repair access and device safety that maximises community wellbeing' ([155] p. 146). The Productivity Commission commented: 'The review should consider whether current regulations create incentives for manufacturers to restrict repair, and examine potential ways to improve repair access for low-risk medical devices or for highly-qualified independent repair technicians' ([155] p. 146).

In a separate inquiry, the Productivity Commission was asked to investigate Australia's vulnerable supply chains in the wake of the COVID-19 crisis. The Productivity Commission reflected that 'Australia's supply chains proved generally resilient in response to the COVID-19 pandemic, unexpected trade restrictions, the devastating 2019–20 bushfires and 2021 floods in Eastern Australia' [160]. However, the Productivity Commission acknowledged that 'these experiences have highlighted potential vulnerabilities in Australia's supply chains', noting that 'the onset of COVID-19 saw immediate impacts on logistics and transport' [160]. The Productivity Commission particularly highlighted issues in respect of access to PPE: 'A global surge in demand and panic buying of some essential goods, notably personal protective equipment, with export restrictions placed on such products by some governments, added a degree of urgency to the unfolding situation' [160]. The Productivity Commission noted: 'Australia was not unique in this respect, with most economies manifesting concerns about how their reliance on imports could jeopardise their ability to meet their population's needs during the COVID-19 pandemic' [160]. The Productivity Commission observed that COVID-19 prompted calls for onshoring-but not everyone agreed.

4.3. The Albanese Government

In 2022 and 2023, the new Albanese Government has provided an initial response to the Productivity Commission inquiry on the right to repair. In 2022, the Assistant Minister for Competition the Hon. Dr. Andrew Leigh MP outlined his initial thoughts about the Productivity Commission report on the right to repair [161]. He recognized that 'Smartphones, watches, fridges, *medical devices*, exercise equipment—our everyday consumer products are increasingly incorporating sophisticated technology' (emphasis added) [161].

Leigh acknowledged: The challenge for consumers globally—not just in Australia—is that technological advances can increase the cost and complexity of repairs' [161]. He argued that the Albanese Government would boost competition in the repair sector 'by ensuring big technology companies cannot create monopolies that allow them to profiteer at the expense of Australians' [161]. Leigh maintained that the Government would protect 'Australians' "right to repair", which gives households and businesses the ability to have their products repaired at a competitive price using a repairer of their choice' [161].

The Albanese Government has thus far focused on the right to repair in motor vehicles [162], and agricultural markets [163]. The Albanese Government has not yet progressed on the spin-off topic of the medical right to repair. As the COVID-19 threat has receded, perhaps the Albanese Government has considered the question of the medical right to repair to be less of a pressing priority. However, it is argued here that the medical right to repair should indeed be a political priority, especially given that a recurrence of the COVID-19 crisis or new pandemics could create similar problems in respect of repairs.

5. The Right to Repair and the TRIPS Waiver

This article has provided national case studies of debates over the right to repair in Italy, the United States, and Australia during the COVID-19 crisis. National law reform initiatives in respect of a medical right to repair also raise larger questions about consistency and conformity with international intellectual property law—most notably, the *TRIPS Agreement* 1994. There is certainly an international dimension to the topic of intellectual property and the right to repair during the COVID-19 crisis. There has been discussion as to whether the proposed *TRIPS Waiver* should extend to the right to repair. Such international debate raises larger questions about the relationship between intellectual property, trade, and the SDGs.

5.1. The TRIPS Agreement 1994

The *TRIPS Agreement* 1994 lays down international standards and norms in respect intellectual property law and policy for WTO members [61].

The right to repair raises questions about the nature and scope of patent law and patent exceptions under the *TRIPS Agreement* 1994. Professor Joshua Sarnoff has explored the international dimensions of patent law and the right to repair during the COVID-19 crisis [164]. He has argued that 'legislative measures to assure the right to repair are fully consistent with the World Trade Organization's [*TRIPS Agreement* 1994]' [164]. Likewise, Muhammad Zaheer Abbas has explored whether a medical right to repair would be consistent with the *TRIPS Agreement* 1994 [158]. He has argued that 'Facilitating increased experimental and repair activity by creating a safe harbor for experimentation with medical devices will better prepare countries to deal with a future pandemic' [158]. Rosa Ballardini and her colleagues have explored what exceptions would be permitted for 3D printing in medical emergencies under the *TRIPS Agreement* 1994 [65].

The right to repair certainly raises issues about the principles and objectives of the *TRIPS Agreement* 1994—as well as provisions on patent rights and patent exceptions. The context of COVID-19 certainly raises further questions about the operation of patent flexibilities during a public health crisis. The *Doha Declaration* 2001 has recognized that nation states are able to make use of intellectual property flexibilities to deal with public health matters [165,166]. The *WTO General Council Decision* 2003 allowed for compulsory licensing to enable the export of pharmaceutical drugs to developing countries, which lacked manufacturing capacity [167].

It is worth noting that the right to repair also raises questions about other domains of intellectual property—including copyright law, designs law, trade mark law, and trade secrets. Sean Flynn and colleagues, for instance, have highlighted that 'Access to copyrighted materials is necessary to create and repair many medical devices needed to treat COVID-19' [168]. Designs law has been invoked in relation to medical designs (including for devices and personal protective equipment). Trade mark law has also raised questions

in respect of the repair of trade marked goods. There has been tension between the right to repair and the protection of confidential information and trade secrets. In the context of COVID-19, the right to repair has raised questions about the use of intellectual property exceptions during a public health crisis. As such, there is a need to contemplate the intellectual property flexibilities permitted by the *TRIPS Agreement* 1994.

5.2. The TRIPS Waiver

The *TRIPS Waiver* featured larger discussions about whether there should be international flexibilities under intellectual property during public health emergencies.

India and South Africa put forward the proposal for a TRIPS Waiver [35]. The proposal involved a waiver from certain provisions of the TRIPS Agreement 1994 for the prevention, containment, and treatment of COVID-19. The proposal considered access to vaccines, therapeutics, medical products, and health equipment. In its original conception, the TRIPS Waiver could have facilitated the right to repair for purposes related to the COVID-19 crisis [35]. India and South Africa emphasized: 'An effective response to COVID-19 pandemic requires rapid access to affordable medical products including diagnostic kits, medical masks, other personal protective equipment and ventilators, as well as vaccines and medicines for the prevention and treatment of patients in dire need' [35]. India and South Africa feared: 'Critical shortages in medical products have also put at grave risk patients suffering from other communicable and non-communicable diseases' [35]. India and South Africa observed: 'To meet the growing supply-demand gap, several countries have initiated domestic production of medical products and/or are modifying existing medical products for the treatment of COVID-19 patients' [35]. India and South Africa commented: 'The rapid scaling up of manufacturing globally is an obvious crucial solution to address the timely availability and affordability of medical products to all countries in need' [35]. India and South Africa were concerned that intellectual property rights were hindering access to medical products: 'There are several reports about intellectual property rights hindering or potentially hindering timely provisioning of affordable medical products to the patients' [35]. So, while in popular discourse the TRIPS Waiver has focused on vaccines, the original proposal was broad enough to embrace a right to repair.

At an international level, the United States Government equivocated as to its position on the topic of the *TRIPS Waiver*. The Trump administration was hostile to the adoption of a *TRIPS Waiver*, as promoted by India and South Africa [169]. In an important shift, the Biden administration announced that it would support a *TRIPS Waiver*—although only for vaccines. The United States Trade Representative (USTR) Katharine Tai commented as follows: 'This is a global health crisis, and the extraordinary circumstances of the COVID-19 pandemic call for extraordinary measures. The Administration believes strongly in intellectual property protections, but in service of ending this pandemic, supports the waiver of those protections for COVID-19 vaccines' [170].

The USTR stressed: 'The Administration's aim is to get as many safe and effective vaccines to as many people as fast as possible' [170]. Katharine Tai promised: 'As our vaccine supply for the American people is secured, the Administration will continue to ramp up its efforts—working with the private sector and all possible partners—to expand vaccine manufacturing and distribution' [170]. Tai also said that the United States Government 'will also work to increase the raw materials needed to produce those vaccines' [170]. The limitation of a *TRIPS Waiver* to vaccines would of course exclude the right to repair for PPE and medical equipment.

The Biden administration was criticized, though, for not being a vocal supporter of the *TRIPS Waiver* during the negotiations.

The Governments of Australia and its neighbor New Zealand largely followed the lead of the United States in the debate over the *TRIPS Waiver*.

After the Biden administration decided to support a *TRIPS Waiver* for vaccines, Australia and New Zealand expressed a willingness to support a *TRIPS Waiver* for vaccines [171,172]. In September 2021, the Coalition Government announced its change in

position [173]. The trade minister Dan Tehan commented: 'When the US came out and said this, the prime minister welcomed that news' [173]. He emphasized: 'We continue to work constructively in Geneva to do everything we can to expand the production of vaccines globally because we need everyone across the globe to get access to a vaccine, ultimately, to be able to be safe' [173]. Tehan maintained: 'We've already expressed that support and we've been working with countries to get a resolution to this issue' [173].

However, in spite of this shift in position of the US and its allies, there remained opposition to a *TRIPS Waiver* for vaccines, or more broadly, from the EU and some of its member states, the United Kingdom, and Switzerland. As a result, the World Trade Organization was unable to reach consensus amongst member states on such a formula for a *TRIPS Waiver*.

5.3. Ministerial Decision

At an international level, it should be observed that the EU opposed the relaxation of intellectual property rights under the *TRIPS Agreement* 1994 during the COVID-19 crisis. The European Union was opposed to a broad *TRIPS Waiver*—much to the despair of the public health community [174,175]. Instead, the EU promoted their own counter-proposal focused on voluntary licensing [176,177].

The EU also promulgated the Quad Proposal [178,179]. The Quad Proposal formed the foundation of what was to become the *Ministerial Decision on the TRIPS Agreement and Public Health* 2022 [37]. The Ministerial Decision was limited in its scope to vaccines. There has been discussion as to whether the Ministerial Decision should be extended to diagnostics and therapeutics [180]. It would seem, though, that the Ministerial Decision would not include the right to repair.

In the end, the World Trade Organization agreed to a much more limited Ministerial Decision [37]. The scope of this declaration is very narrowly limited to compulsory licensing for export of vaccines. USTR Katharine Tai was upbeat about the disappointing decision, saying 'The text-based negotiations with other WTO Members that we called for have produced accommodations to the intellectual property rules for COVID-19 vaccines that can facilitate a global health recovery' [181]. She commented: 'Through difficult and protracted discussions, Members were able to bridge differences and achieve a concrete and meaningful outcome to get more safe and effective vaccines to those who need it most' [181]. As Tai noted, 'This agreement shows that we can work together to make the WTO more relevant to the needs of regular people' [181]. She stressed: 'During a global pandemic, under difficult circumstances, the WTO moved quickly to address a major global challenge and respond to the strong desire of our African partners to produce a meaningful outcome' [181]. Tai promised: 'Going forward, the Biden Administration will continue work with WTO Members, the private sector, and other partners to expand vaccine manufacturing and distribution to facilitate the global health recovery needed for a robust global economic recovery' [181].

There has been some debate as to whether the clause should also be extended to diagnostics and therapeutics [182]. The United States International Trade Commission is holding hearings to investigate COVID-19 diagnostics and therapeutics and flexibilities under the *TRIPS Agreement* 1994 [183].

In any case, the scope of the Ministerial Decision is far too narrow to deal with the right to repair of medical equipment and 3D printing PPE and other similar supplies. Arguably, there is a disjuncture between the Biden Administration's support for a right to repair at a domestic level and its failure to support a right to repair during the discussions over the *TRIPS Waiver*.

The new Australian Labor Party Government led by Anthony Albanese welcomed the Ministerial Conference decision [184]. The Minister for Trade and Tourism Senator Don Farrell commented: 'In the first Ministerial Conference since the COVID-19 pandemic, WTO Members decided on a waiver of certain intellectual property rules to promote access to COVID-19 vaccines—sending a powerful signal of global solidarity in fighting COVID-19 and reaffirming the role the WTO can play in responding to urgent global issues' [184]. He observed that the decision will 'support equitable access to vaccines for developing countries, including across the Indo-Pacific region' [184]. However, this statement really failed to contemplate the very limited circumstances under which the Ministerial Decision could be invoked.

Moreover, the Ministerial Decision has no application to the circumstances covered in the scenario of 3D printing PPE and other medical equipment. There is a lack of international consideration of intellectual property and the right to repair in circumstances of public health emergencies. It has been argued in this article that should be recognition of a medical right to repair, not only in national laws and policies, but at an international level. Such a recognition would help bolster a COVID-19 recovery and future pandemic responses—especially in light of the UN SDGs.

6. Conclusions

To recap, this article has explored a number of themes, including the utility of 3D printing and additive manufacturing in the delivery of PPE and medical supplies; the role of the Maker Movement in responding to public health emergencies; the role of open licensing; intellectual property reform, particularly relating to the right to repair; and the geopolitical debate about whether there should be a *TRIPS Waiver*, and, if so, what its scope should be. It has argued that the recognition of a right to repair (including a medical right to repair) will promote public health, innovation, and sustainable development.

This article has highlighted the deployment of 3D printing and additive manufacturing during the COVID-19 crisis to address shortages in PPE and medical equipment. It has documented how 3D printing and additive manufacturing was used during the public health emergency in several jurisdictions, such as the European Union, the United States, and Australia. This work has highlighted the utility and flexibility of 3D printing and additive manufacturing in respect of healthcare and medicine, public health emergencies, and humanitarian aid. An article in *Nature Reviews Materials* observed: 'In the heat of the COVID-19 pandemic, 3D printing has stepped up to become a vital technology to support improved healthcare and our general response to the emergency' [185]. The piece observed: 'The digital versatility and quick prototyping of 3D printing empowers a swift mobilization of the technology and hence a rapid response to emergencies' [185]. The review suggested that 3D printing could be useful in the future in dealing with SDGs and challenges: 'The crisis has highlighted how 3D printing can be at the base of a greener and more environmentally friendly future' [185].

There has been a strong connection between the right to repair and the UN SDGs especially responsible production and consumption (SDG 12), but also good health and well-being (SDG 3), innovation (SDG 9), and partnerships for the goals (SDG 17). The United Nations Development Programme has sought to make greater use of 3D printing and additive manufacturing in its accelerator labs to boost the UN SDGs [186]. As a technology, 3D printing could certainly enable the achievement of the UN SDGs—in particular, the goal relating to sustainable production and consumption; the goal associated with innovation; the goal focused upon public health; and the goal focused on partnerships [187].

This article has highlighted how the Maker Movement's use of 3D printing during the COVID-19 crisis was facilitated by patent pledges and open licensing. In the European Union, open licensing could have averted some of the concerns about intellectual property infringement. In the United States, a number of 3D printing initiatives during the COVID-19 crisis took up the Open COVID Pledge [90]. In Australia, there was adoption of openlicensing models in terms of developing PPE. Likewise, South Africa made use of openlicensed designs for PPE and medical equipment. Professor Joshua Pearce observed that there is a need to provide better protection of open-source developers: 'Lawyers can help as well as technologists by developing Good-Samaritan laws to protect makers, designers, and users of open medical hardware, as well as to compel those with knowledge that will save lives to share it' [91]. Pearce concluded: 'Requiring all citizen-funded research to be released with free and open source licenses in the future will prevent such artificial scarcity from needlessly allowing people to die' [91]. At an international level, UNESCO and others have promoted open science models to foster collaboration during the COVID-19 crisis [188,189]. The COVID-19 crisis has underlined the need for better governance of the medical commons [190].

This article has also highlighted that intellectual property can provide barriers and obstacles to the rapid manufacture of PPE and other medical equipment. There have been a number of controversies during the COVID-19 crisis, in which there have been concerns about intellectual property imposing restrictions on repair. Professor Joshua Pearce has been concerned that a key inhibitor to the rapid manufacture of ventilators, other medical equipment, and goods in general has been intellectual property [191]. This article has called for law reform on intellectual property and the right to repair at a national level—particularly in light of the COVID-19 crisis. There have been various law reform proposals in the key jurisdictions studied in this article, but they have not yet necessarily come to full fruition. There have been concerns as well that such proposals have been too narrowly framed and circumscribed. In the European Union, there has been policy movement towards a stronger recognition towards a right to repair—although from the perspective of environmental stewardship and sustainable development. In the United States, there was a concrete proposal for the Critical Medical Infrastructure Right-to-Repair Act of 2020 (US) put forward by Senator Ron Wyden. While this was a comprehensive proposal, it has not yet been taken up by the United States Congress. In Australia, the Productivity Commission has made a number of recommendations to better recognize the right to repair in Australia. However, the law reform body has hesitated about a medical right to repair—calling for a new, separate inquiry on that particular topic. As part of their flexibilities under the TRIPS Agreement 1994, national governments do need to implement broad-based defenses in respect to the right to repair—across the various regimes of intellectual property.

This article has also highlighted the need for a rethink of international trade law in its approach to intellectual property and public health. A broad-based *TRIPS Waiver*—as envisaged by the Governments of India and South Africa—would have helped provide protection for the right to repair during the COVID-19 crisis [35]. A narrower *TRIPS Waiver* limited to vaccines or vaccines, treatments, and diagnostics would not have included the right to repair [170]. The *Ministerial Decision on the TRIPS Agreement* 2022 certainly does not embrace the right to repair [37]. The Ministerial Decision does not deal adequately with the problem of access to affordable medical equipment and supplies—which has been a major issue during the COVID-19 crisis.

In addition to matters of intellectual property and the right to repair, there has also been questions of medical regulation of 3D printing and additive manufacturing during public health emergencies [192]. This article has touched upon some of those public policy issues, but they have not been the central focus of the piece. There has certainly been significant discussion of regulation of health-related 3D printing, medical 3D printing, and bioprinting [193]. There have also been some special protocols in relation to the use of such technology during the COVID-19 crisis [194–196]. Professor Joshua Pearce contends: 'Policies are needed to protect the productivity of laboratories, makerspaces, and fabrication facilities during a pandemic to enable such products to be fabricated when the need arises' [91]. Pearce acknowledges: 'These products need to be safe, but there is also a need for streamlining the regulatory process' [91]. This topic is beyond the scope of this paper, but it deserves further research and examination. Likewise, there have been issues arising in respect of product liability in respect of the use of 3D printing and additive manufacturing generally [197–199]. There are perhaps special considerations of product liability of 3D printing and additive manufacturing during public health emergencies [98,200-202]. More generally, there has been a discussion about product liability and medical indemnity issues in respect of other medical products, such as vaccines, treatments, and diagnostics during the COVID-19 crisis [30]. Such matters of 3D printing, medical regulation, and product

liability during a public health emergency certainly deserve further consideration and analysis in the future.

Funding: Australian Research Council Discovery Grant on 'Inventing the Future: Intellectual Property and 3D Printing' (DP170100758).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- He, S.; Ojo, A.; Beckman, A.L.; Gondi, S.; Gondi, S.; Betz, M.; Faust, J.S.; Choo, E.; Kass, D.; Raja, A.S. The Story of #GetMePPE and GetUsPPE.org to Mobilize Health Care Response to COVID-19: Rapidly Deploying Digital Tools for Better Health Care. *J. Med. Internet Res.* 2020, 22, e20469. Available online: https://www.jmir.org/2020/7/e20469 (accessed on 1 August 2023).
- 2. Knaus, C. Manufacturing Face Shields within Days: How Australian Industry is Pivoting to Fight Coronavirus. *The Guardian*, 5 April 2020. Available online: https://www.theguardian.com/world/2020/apr/05/manufacturing-face-shields-mat-bowtell-3d-printing-australia-fight-coronavirus (accessed on 1 August 2023).
- Radfar, P.; Bazaz, S.R.; Mirakhorli, F.; Warkiani, M.E. The Role of 3D Printing in the Fight against COVID-19 Outbreak. J. 3D Print. Med. 2021, 5, 51–60. Available online: https://www.futuremedicine.com/doi/10.2217/3dp-2020-0028 (accessed on 1 August 2023). [CrossRef]
- Applin, S. Maker Culture's DIY Spirit is Helping Us Get Through this Pandemic. *Fast Company*, 15 April 2020. Available online: https://fastcompany.com/90489974/the-maker-faire-spirit-is-helping-the-world-tackle-the-covid-19-crisis (accessed on 1 August 2023).
- Doctorow, C. Right to Repair in Times of Pandemic, Electronic Frontier Foundation, 19 March 2020. Available online: https://eff.org/ deeplinks/2020/03/right-repair-times-pandemic (accessed on 1 August 2023).
- 6. Giblin, R.; Doctorow, C. Chokepoint Capitalism: How Big Tech and Big Content Captured Creative Labor Markets and How Well's Win Them Back; Penguin Random House: New York, NY, USA; London, UK, 2022.
- 7. Drahos, P.; Mayne, R. (Eds.) *Global Intellectual Property Rights: Knowledge, Access, and Development*; Palgrave Macmillan: Basingstoke, UK; New York, NY, USA, 2003.
- 8. Ho, C. Access to Medicine in the Global Economy: International Agreements on Patents and Related Rights; Oxford University Press: Oxford, UK, 2011.
- 9. Kilic, B. Boosting Pharmaceutical Innovation in the Post-TRIPS Era: Real-Life Lessons for the Developing World; Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2014.
- 10. 't Hoen, E. Private Patents and Public Health: Changing Intellectual Property Rules for Access to Medicines; AMB Publishers: Amsterdam, The Netherlands, 2016.
- 11. Ragavan, S.; Vanni, A. (Eds.) Intellectual Property Law and Access to Medicines: TRIPS Agreement, Health and Pharmaceuticals; Routledge: Abingdon, UK, 2021.
- 12. Correa, C.; Hilty, R. (Eds.) Access to Medicines and Vaccines: Implementing Flexibilities Under Intellectual Property Law; Springer: Cham, Switzerland, 2021.
- 13. Santos, A. Vaccines as Technology: Innovation, Barriers, and the Public Health; Cambridge University Press: Cambridge, UK, 2022.
- 14. Heath, C.; Sanders, A.K. (Eds.) *Spares, Repairs and Intellectual Property Rights;* Kluwer Law International: Alphen aan den Rijn, The Netherlands, 2009.
- 15. Perzanowski, A.; Schultz, J. The End of Ownership: Personal Property in the Digital Economy; MIT Press: Cambridge, MA, USA, 2016.
- 16. Wackman, J.; Knight, E. Repair Revolution: How Fixers are Transforming our Throwaway Culture; New World Library: Novato, CA, USA, 2020.
- 17. Perzanowski, A. The Right to Repair: Reclaiming the Things We Own; Cambridge University Press: Cambridge, UK, 2022.
- 18. Weinberg, M. It Will Be Awesome If They Don't Screw It up: 3D Printing, Intellectual Property, and the Fight over the Next Great Disruptive Technology; Public Knowledge: Washington, DC, USA, 2012.
- 19. Hornick, J. 3D Printing Will Rock the World; Createspace: North Charleston, SC, USA, 2015.
- 20. Daly, A. Socio-Legal Aspects of the 3D Printing Revolution; Palgrave Macmillan: London, UK, 2016.
- 21. Mendis, D.; Lemley, L.; Rimmer, M. (Eds.) *3D Printing and Beyond: Intellectual Property and Regulation*; Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2019.
- 22. Osborn, L. 3D Printing and Intellectual Property; Cambridge University Press: Cambridge, UK, 2019.
- 23. Griffin, J. *The State of Creativity: The Future of 3D Printing, 4D Printing and Augmented Reality;* Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2019.
- 24. Chan, H.K.; Choo, H.L.; Osuji, O.; Griffin, J. (Eds.) Intellectual Property Rights And Emerging Technology: 3D Printing in China; Routledge: Abingdon, UK, 2019.

- 25. Flood, C.M.; MacDonnell, V.; Philpott, J. Vulnerable: The Law, Policy and Ethics of COVID-19; The University of Ottawa Press: Ottawa, ON, Canada, 2020.
- 26. Gans, J. The Pandemic Information Gap: The Brutal Economics of COVID-19; The MIT Press: Cambridge, MA, USA, 2020.
- 27. Ramraj, V. (Ed.) COVID-19 in Asia: Law and Policy Contexts; Oxford University Press: Oxford, UK, 2021.
- 28. Bennett, B.; Freckelton, I. (Eds.) *Pandemics, Public Health Emergencies and Government Powers: Perspectives on Australian Law;* Federation Press: Sydney, Australia, 2021.
- 29. The Independent Panel for Pandemic Preparedness and Response. *COVID-19: Make It the Last Pandemic;* World Health Organization: Geneva, Switzerland, 2021.
- 30. Bennett, B.; Freckelton, I.; Wolf, G. (Eds.) COVID-19, Law and Regulation: Rights, Freedoms, and Obligations in a Pandemic; Oxford University Press: Oxford, UK, 2022.
- Cimoli, M.; Dosi, G.; Maskus, K.E.; Okediji, R.L.; Reichman, J.H.; Stiglitz, J.E. (Eds.) Intellectual Property Rights: Legal and Economic Challenges for Development; Oxford University Press: Oxford, UK, 2014.
- 32. Chon, M.; Roffe, P.; Abdel-Latif, A. (Eds.) *The Cambridge Handbook on Public-Private Partnerships, Intellectual Property Governance, and Sustainable Development*; Cambridge University Press: Cambridge, UK, 2018.
- Rognstad, O.-A.; Ørstavik, I. (Eds.) Intellectual Property and Sustainable Markets; Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2021.
- Pihlajarinne, T.; Mahonen, J.; Upreti, P. (Eds.) Intellectual Property Rights in the Post Pandemic World: An Integrated Framework of Sustainability, Innovation and Global Justice; Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2023.
- 35. India and South Africa. Waiver from Certain Provisions of the TRIPS Agreement for the Prevention, Containment and Treatment of COVID-19, IP/C/W/669 (TRIPS Waiver).
- 36. Patnaik, P. *The TRIPS Waiver Negotiations at the World Trade Organization: A Reportage from Geneva Health Files;* Geneva Health Files: Geneva, Switzerland, 2022.
- World Trade Organization. *Ministerial Decision on the TRIPS Agreement;* WTO Doc WT/MIN(22)/30 WT/L/1141 (Ministerial Decision 2022).
- Mitchell, A.; Taubman, A.; Samlidis, T. The Legal Character and Practical Implementation of a TRIPS Waiver for COVID-19 Vaccines. Fordham Intellect. Prop. Media Entertain. Law J. 2022, 33, 100–175. [CrossRef]
- 39. The European Parliament Has Acknowledged that '3D Printing Has an Enormous Potential to Transform Supply Chains in Manufacturing Which Could Help EUROPE Increase Output Levels' and That 'the Application of This Technology Offers New Opportunities for Business Development and Innovation': European Parliament, 'Three-Dimensional Printing: Intellectual Property Rights and Civil Liability', Text adopted 3 July 2018, P8_TA(2018)0274. Available online: https://www.europarl.europa.eu/doceo/ document/TA-8-2018-0274_EN.html?redirect (accessed on 1 August 2023).
- European Patent Office. Patents and Additive Manufacturing: Trends in 3D Printing Technologies; European Patent Office: Munich, Germany, 2020. Available online: http://documents.epo.org/projects/babylon/eponet.nsf/0/C2F0871212671851C125859F0040BCCA/ \$FILE/additive_manufacturing_study_en.pdf (accessed on 1 August 2023).
- Mendis, D.; Nordemann, J.; Ballardini, R.; Brorsen, H.; Calatrava Moreno, M.D.C.; Robson, J.; Dickens, P. *The Intellectual Property Implications of the Development of Industrial 3D Printing*; European Commission: Brussels, Belgium, 2020. Available online: http://eprints.bournemouth.ac.uk/33718/ (accessed on 1 August 2023).
- 42. Davies, S.R. Hackerspaces: Making the Maker Movement; Polity Press: Cambridge, UK, 2017.
- Fab Lab Foundation. The Fab Lab Network's Response to COVID-19. Available online: https://fabfoundation.org/Covid19-Blast (accessed on 1 August 2023).
- 44. Fab Labs. Available online: https://www.fablabs.io/ (accessed on 1 August 2023).
- 45. Bull, M. The Italian Government Response to COVID-19 and the Making of a Prime Minister. *Contemp. Ital. Politics* **2021**, *13*, 149–165. [CrossRef]
- Bosa, I.; Castelli, A.; Castelli, M.; Ciani, O.; Compagni, A.; Galizzi, M.M.; Ghislandi, S.; Giannoni, M.; Marini, G.; Vainieri, M. Response to COVID-19: Was Italy (Un)prepared? *Health Econ. Policy Law* 2021, 17, 1–13. Available online: https://www.ncbi.nlm.nih.gov/pmc/ articles/PMC7985656/ (accessed on 1 August 2023).
- Sanfield, M. The Italian Response to the COVID-19 Crisis: Lessons Learned and Future Direction in Social Development. Int. J. Community Soc. Dev. 2020, 2, 191–210.
- Brown, F. Firm "Refuses to Give Blueprint" For Coronavirus Equipment that Could Save Lives. *Metro*, 16 March 2020. Available online: https://metro.co.uk/2020/03/16/firm-refuses-give-blueprint-coronavirus-equipment-save-lives-12403815/ (accessed on 1 August 2023).
- Sher, D. Italian Hospital saves COVID-19 Patient's Lives by 3D Printing Valves for Reanimation Devices. 3D Printing Media, 14 March 2020. Available online: https://www.3dprintingmedia.network/covid-19-3d-printed-valve-for-reanimation-device/ (accessed on 1 August 2023).
- Peters, J. Medical Company Threatens to Sue Volunteers that 3D–printed Valves for Life-Saving Coronavirus Treatments. *The Verge*, 17 March 2020. Available online: https://theverge.com/2020/3/17/21184308/coronavirus-italy-medical-company-threatenssue-3d-print-valves-treatments (accessed on 1 August 2023).
- Moody, G. Volunteers 3D-Print Unobtainable \$11,000 Valve For \$1 To Keep COVID-19 Patients Alive; Original Manufacturer Threatens To Sue. *TechDirt*, 17 March 2020. Available online: https://www.techdirt.com/articles/20200317/04381644114

/volunteers-3d-print-unobtainable-11000-valve-1-to-keep-covid-19-patients-alive-original-manufacturer-threatens-to-sue. shtml (accessed on 1 August 2023).

- 52. Feldman, A. Meet The Italian Engineers 3D-Printing Respirator Parts For Free To Help Keep Coronavirus Patients Alive. Forbes, 19 March 2020. Available online: https://www.forbes.com/sites/amyfeldman/2020/03/19/talking-with-the-italian-engineerswho-3d-printed-respirator-parts-for-hospitals-with-coronavirus-patients-for-free/?sh=2a668678f1a5 (accessed on 1 August 2023).
- Serra, M. Coronavirus, 'Are the Valves Missing in Resuscitation? Let's Make Them with the 3D Printer'. *La Stampa*, 14 March 2020. Available online: https://www.lastampa.it/cronaca/2020/03/14/news/coronavirus-mancano-le-valvole-in-rianimazione-facciamole-con-la-stampante-3d-1.38593309/ (accessed on 1 August 2023).
- 54. Urian, B. Maker of \$11,000 Ventilator Valves Threatens to Sue Volunteers Using \$1 3D-Printed Replicas That Just Saved 10 Coronavirus Patients! *Tech Times*, 17 March 2020. Available online: https://www.techtimes.com/articles/248121/20200317/ maker-ventilator-valves-threatens-sue-volunteers-using-3d-printed-coronavirus.htm (accessed on 1 August 2023).
- 55. Randall, I. 3D Printer Saves Ten Italian Coronavirus Patients' Lives by Producing a Replacement Valve for a Broken Ventilator in just a Few Hours after the Hospital's Supplier Falls Short. *Mail Online*, 17 March 2020. Available online: https://www.dailymail.co.uk/sciencetech/article-8117399/3D-printer-saves-Italian-coronavirus-patients-lives-making-new-valve-broken-ventilator.html (accessed on 1 August 2023).
- 56. Garrett, A. Medical Company Threatens to Sue Volunteers for 3D-Printing Valves to Treat Coronavirus Patients. *Distractify*, 18 March 2020. Available online: https://www.distractify.com/p/medical-valves-sue-3d-printing (accessed on 1 August 2023).
- Toussaint, K. These Good Samaritans with a 3D Printer are Saving Lives by Making New Respirator Valves for Free. *Fast Company*, 16 March 2020. Available online: https://www.fastcompany.com/90477940/these-good-samaritans-with-a-3d-printer-aresaving-lives-by-making-new-respirator-valves-for-free (accessed on 1 August 2023).
- Sparrow, N. Updated: OEM Threatens to Sue Startup that Saved Lives by 3D Printing Medical Valves. *Plastics Today*, 19 March 2020. Available online: https://www.designnews.com/medical/updated-oem-threatens-sue-startup-saved-lives-3d-printing-medical-valves (accessed on 1 August 2023).
- Atienza, E. Fact Check: Did an Italian Company Threaten to Sue Two Men for 3D Printing Valves for a Respiratory Device? *Check Your Fact*, 31 March 2020. Available online: https://checkyourfact.com/2020/03/31/fact-check-intersurgical-threaten-sue-men-3d-printing-valves-coronavirus-italy/ (accessed on 1 August 2023).
- 60. Osborn, L. 3D Printing, Patent Infringement, and the Coronavirus. *Patently-O*, 19 March 2020. Available online: https://patentlyo.com/patent/2020/03/printing-infringement-coronavirus.html (accessed on 1 August 2023).
- Agreement on Trade-related Aspects of Intellectual Property Rights ('TRIPS Agreement'), adopted 15 April 1994 at Marrakesh, TRT/WTO01/001, entry into force 1 January 1995 (annex 1C of the Marrakesh Agreement Establishing the World Trade Organization, 1867 UNTS 3).
- Contreras, J. Research and Repair: Expanding Exceptions to Patent Infringement in Response to a Pandemic. J. Law Biosci. 2020, 7, Isaa014. Available online: https://academic.oup.com/jlb/article/7/1/lsaa014/5828395?login=true (accessed on 1 August 2023). [CrossRef]
- 63. Moody, G. Walled Culture: How Big Content Uses Technology and the Law to Lock Down Culture and Keep Creators Poor; BTF Press: Concord, MA, USA, 2022.
- Di Santo, G.P.; Girardello, G.; Cucinella, G. Insights on Pharmaceuticals Patents, and COVID-19 from an Italian Perspective'. *IP Kitten*, 22 April 2020. Available online: https://ipkitten.blogspot.com/2020/04/guest-post-insights-on-pharmaceuticals.html (accessed on 1 August 2023).
- 65. Ballardini, R.M.; Mimler, M.; Minssen, T.; Salmi, M. 3D Printing, Intellectual Property Rights and Medical Emergencies: In Search of New Flexibilities. *IIC—Int. Rev. Intellect. Prop. Compet. Law* 2022, *53*, 1149–1173. [CrossRef]
- 66. Pihlajarinne, T. Paving the Way for the Environment: Channelling 'Strong' Sustainability into the European IP System. *Eur. Intellect. Prop. Rev.* **2020**, *42*, 239–250.
- 67. Pihlajarinne, T. European Steps to the Right to Repair: Towards a Comprehensive Approach to a Sustainable Lifespan of Products and Materials? *Aust. Intellect. Prop. J.* **2020**, *31*, 111–119. [CrossRef]
- Pihlajarinne, T. Repairing and Re-using from an Exclusive Rights Perspective: Towards Sustainable Lifespan as Part of a New Normal? In *Intellectual Property and Sustainable Markets*; Rognstad, O.-A., Ørstavik, I., Eds.; Edward Elgar: Cheltenham, UK; Northampton, MA, USA, 2021; pp. 81–100.
- Mikolajczak, C. Why Repair is Essential in Times of Crisis. *Right to Repair*, 21 April 2020. Available online: https://repair.eu/ news/why-repair-is-essential-in-times-of-crisis/ (accessed on 1 August 2023).
- Materialise, N.V. COVID-19 Initiatives. *Materialise*. Available online: https://www.materialise.com/en/about/impact/people/ covid-19-response (accessed on 1 August 2023).
- 71. Materialise, N.V. How 3D Printing at the Point-of-Care Can Support in Fighting COVID-19. *Materialise*. Available online: https://www.materialise.com/en/inspiration/articles/point-of-care-3d-printing-covid-19 (accessed on 1 August 2023).
- Shapeways. How 3d Printing is Contributing to COVID 19 Relief. *Shapeways Blog*, 22 April 2020. Available online: https://www.shapeways.com/blog/archives/40281-how-3d-printing-is-contributing-to-covid-19-relief.html> (accessed on 1 August 2023).

- 73. Carlota, V. Prusa Mass Produces Medical Shields to Fight COVID-19. *3D Natives*, 20 March 2020. Available online: https://www.3dnatives. com/en/prusa-medical-shields-covid-19-200320205/# (accessed on 1 August 2023).
- 74. Prusa Research, 3D Printed Face Shields For Medics and Professionals, Prusa Research. Available online: https://www.prusa3d. com/page/covid-19_379/ (accessed on 1 August 2023).
- 75. Prusament, Case Study: Prusament is Helping Fight COVID-19. *Prusament* 3 January 2022. Available online: https://prusament.com/case-study-prusament-for-assistance-against-covid-19/ (accessed on 1 August 2023).
- 76. Goehrke, S. Making Made Right: This Czech Company Guides Global 3-D Printing Pandemic Response. *Forbes*, 28 March 2020. Available online: https://www.forbes.com/sites/sarahgoehrke/2020/03/28/making-made-right-this-czech-company-guides-global-3-d-printing-pandemic-response/?sh=2455448a624b (accessed on 1 August 2023).
- CECIMO. Industrial Innovation after COVID-19: Four Points to Support AM's Long-Term Growth. CECIMO, 25 May 2020. Available online: https://www.cecimo.eu/publications/industrial-innovation-after-covid-19-four-points-to-support-ams-long-term-growth/ (accessed on 1 August 2023).
- World Health Organization. COVID-19, United States of America Situation. 2023. Available online: https://covid19.who.int/region/amro/country/us (accessed on 1 August 2023).
- American Manufacturing Crisis Production Response. America Makes COVID-19 Response: Assessing the Role of Additive Manufacturing in Support of the U.S. COVID 199 Response; America Makes: Youngstown, OH, USA, 2021. Available online: https://www.fda.gov/emergency-preparedness-and-response/ocet-advanced-manufacturing/assessing-role-additivemanufacturing-support-us-covid-19-response (accessed on 1 August 2023).
- Luna, T.; Lin, R.-G.L.A., II. County got 170 Broken Ventilators from Feds; Silicon Valley is Fixing Them, Newsom says. Los Angeles Times, 29 March 2020. Available online: https://www.latimes.com/california/story/2020-03-29/california-broken-ventilatorssilicon-valley-newsom (accessed on 1 August 2023).
- Dougherty, D.; Hwang, V. What's "Plan C" for COVID-19? *Make Magazine*, 22 March 2020. Available online: https://makezine. com/article/maker-news/whats-plan-c-for-covid-19/ (accessed on 1 August 2023).
- Dougherty, D.; Ryan, T. We Need a New CCC for Recovery—The Civic Response Corps. *Make Magazine*, 4 April 2020. Available online: https://makezine.com/2020/04/04/we-need-a-new-ccc-for-recovery-the-civic-response-corps/ (accessed on 1 August 2023).
- 83. HP. 3D Printing in Support of COVID-19 Containment Efforts. *HP*. Available online: https://reinvent.hp.com/us-en-3dprint-COVID-19-containment-applications (accessed on 1 August 2023).
- 84. Chan, R. Electronics Manufacturing Company Adafruit has seen a 90% Plunge in Sales during the Coronavirus Pandemic, but It's Been Able to Overhaul its Operations to Build Medical Supplies for Crisis Relief. *Business Insider*, 26 April 2020. Available online: https://www.businessinsider.com.au/adafruit-builds-ppe-medical-devices-coronavirus-crisis-2020-4?r=US&IR=T (accessed on 1 August 2023).
- 85. Ehrhardt, M. Electronics Maker Adafruit Switches to Producing Face Shields, Other PPE. *Tom's Hardware*, 13 April 2020. Available online: https://www.tomshardware.com/news/adafruit-making-ppe (accessed on 1 August 2023).
- Reichental, A. 3D Printing Responses to COVID19—A Stocktaking. Nexa 3D, 9 July 2020. Available online: https://linkedin.com/ pulse/many-efforts-worked-during-covid-others-failed-heres-what-reichental/ (accessed on 1 August 2023).
- University of Miami. University of Miami Researchers use 3D Printing to Develop a Reusable N95 mask. Press Release, University of Miami, 16 July 2020. Available online: https://news.miami.edu/stories/2020/07/researchers-use-3d-printing-to-develop-areusable-n95-mask.html (accessed on 1 August 2023).
- 88. Open COVID Pledge. Available online: https://opencovidpledge.org/ (accessed on 1 August 2023).
- Open COVID Pledge. NJIT Makes Novel Test Swab Technology Available via the Open COVID Pledge. Open COVID Pledge, 4 May 2020. Available online: https://opencovidpledge.org/2020/05/04/njit-makes-novel-test-swab-technology-available-viathe-open-covid-pledge/ (accessed on 1 August 2023).
- Contreras, J.L.; Eisen, M.; Ganz, A.; Lemley, M.; Molloy, J.; Peters, D.M.; Tietze, F. Pledging Intellectual Property for COVID-19. Nat. Biotechnol. 2020, 38, 1146–1149. [CrossRef]
- 91. Pearce, J. Distributed Manufacturing of Open-Source Medical Hardware for Pandemics. *J. Manuf. Mater. Process.* **2020**, *4*, 49. Available online: https://www.mdpi.com/2504-4494/4/2/49 (accessed on 1 August 2023). [CrossRef]
- FormLabs. Patents for Humanity 2023: 3D Printing COVID-19 Swabs. *FormLabs*, 22 February 2023. Available online: https://formlabs.com/blog/patents-for-humanity-3d-printing/ (accessed on 1 August 2023).
- United States Patent and Trademark Office. Patents for Humanity: COVID-19 Category Award Recipients, United States Patent and Trademark Office. 2020. Available online: https://www.uspto.gov/ip-policy/patent-policy/patents-humanity/patentshumanity-covid-19-category-award-recipients (accessed on 1 August 2023).
- 94. Rimmer, M. Patents for Humanity. WIPO J. 2012, 3, 196–221. [CrossRef]
- 95. United States Government. Memorandum of Understanding: Rapid Response to COVID-19 Using 3D Printing between National Institutes of Health within U.S. Department of Health and Human Services and Food and Drug Administration, U.S. Department of Health and Human Services and Veterans Health Administration within the U.S. Department of Veterans Affairs, 26 March 2020. Available online: https://www.fda.gov/about-fda/domestic-mous/mou-225-20-008 (accessed on 1 August 2023).

- Vincent, B. FDA, NIH, VA Partner to Accelerate 3D-Printed Protective Gear for COVID-19 Response. *NextGov*, 31 March 2020. Available online: https://nextgov.com/emerging-tech/2020/03/fda-nih-va-partner-accelerate-3d-printed-protective-gear-covid-19-response/164246/ (accessed on 1 August 2023).
- McCarthy, M.C.; Di Prima, M.A.; Cruz, P.; Ribic, B.; Wilczynski, J.; Ripley, B.A.; Coburn, J.C. Trust in the Time of COVID-19: 3D Printing and Additive Manufacturing (3DP/AM) as a Solution to Supply Chain Gaps. *NEJM Catalyst*, 10 November 2021. Available online: https://catalyst.nejm.org/doi/full/10.1056/CAT.21.0321 (accessed on 1 August 2023).
- 98. Bultman, M. 3D Printing Ingenuity during Coronavirus Comes With IP Risks. *Bloomberg*, 1 April 2020. Available online: https://news.bloomberglaw.com/health-law-and-business/3d-printing-ingenuity-during-caronavirus-comes-with-ip-risks (accessed on 1 August 2023).
- Morgenstein, M.; Proctor, N. 43,000 Call on Ventilator Manufacturers to Release Repair Information'. *MoPIRG*, 3 April 2020. Available online: https://mopirg.org/news/usp/43000-call-ventilator-manufacturers-release-repair-information (accessed on 1 August 2023).
- 100. Wyden, R. Wyden and Clarke Introduce Bill to Eliminate Barriers to Fixing Critical Medical Equipment During the Pandemic, Press Release. United States Congress. 6 August 2020. Available online: https://www.wyden.senate.gov/news/press-releases/ wyden-and-clarke-introduce-bill-to-eliminate-barriers-to-fixing-critical-medical-equipment-during-the-pandemic- (accessed on 1 August 2023).
- 101. Burke, K. COVID-19 Highlights Why IP Shouldn't Limit the Right to Repair. *Public Knowledge*, 22 May 2020. Available online: https://www.publicknowledge.org/blog/covid-19-highlights-why-ip-shouldnt-limit-the-right-to-repair/ (accessed on 1 August 2023).
- 102. Stella, S. Public Knowledge Applauds Bill Protecting the Public's Right To Repair During Health Crisis, Press Release. Public Knowledge, 6 August 2020. Available online: https://www.publicknowledge.org/press-release/public-knowledge-applauds-bill-protecting-the-publics-right-to-repair-during-health-crisis/ (accessed on 1 August 2023).
- 103. Gault, M.; Koebler, J. Congress Will Consider National Right-to-Repair Legislation for Medical Equipment'. Vice, 6 August 2020. Available online: https://www.vice.com/en/article/akzyy5/congress-will-consider-national-right-to-repair-legislation-formedical-equipment (accessed on 1 August 2023).
- 104. Critical Medical Infrastructure Right-to-Repair Act of 2020 (US) H.R. 7956 Introduced 7 August 2020. Available online: https://www.congress.gov/bill/116th-congress/house-bill/7956 (accessed on 1 August 2023).
- 105. Critical Medical Infrastructure Right-to-Repair Act of 2020 (US) S. 4473 introduced 6 August 2020. Available online: https://www.congress.gov/bill/116th-congress/senate-bill/4473 (accessed on 1 August 2023).
- 106. Paben, J. OEM and iFixit Engage in Copyright Law Spat. E-Scrap News, 18 June 2020. Available online: https://resource-recycling. com/e-scrap/2020/06/18/oem-and-ifixit-engage-in-copyright-law-spat/ (accessed on 1 August 2023).
- 107. Electronic Frontier Foundation. Letter from EFF to Steris on Behalf of iFixit, Electronic Frontier Foundation. 26 May 2020. Available online: https://www.eff.org/document/letter-eff-steris-behalf-ifixit-5-26-2020 (accessed on 1 August 2023).
- Hatchette Book Group. Inc. v. Internet Archive (US District Court for the Southern District of New York, 2023) 20-cv-4160. Available online: https://law.justia.com/cases/federal/district-courts/new-york/nysdce/1:2020cv04160/537900/188/ (accessed on 1 August 2023).
- Wyden, R.; Kullolli, I. Hospitals Need to Be Able to Repair Their Own Medical Equipment. *Future Tense, Slate,* 12 October 2020. Available online: https://slate.com/technology/2020/10/critical-medical-infrastructure-right-to-repair-act-ventilators.html (accessed on 1 August 2023).
- O'Reilly, K.; Morgenstein, M.; Proctor, N. New Senate Right to Repair Bill to Reduce Barriers to Fixing Medical Equipment Including Ventilators. PIRG, 6 August 2020. Available online: https://pirg.org/media-center/new-senate-right-to-repair-bill-toreduce-barriers-to-fixing-medical-equipment-including-ventilators/ (accessed on 1 August 2023).
- Brimhall, C. R Street Applauds Introduction of the Critical Medical Infrastructure Right-to-Repair Act. R Street, 6 August 2020. Available online: https://rstreet.org/2020/08/06/r-street-applauds-introduction-of-the-critical-medical-infrastructure-right-to-repair-act/ (accessed on 1 August 2023).
- Niskanen Center. Niskanen Center Proud to Endorse Critical Medical Infrastructure Right-to-Repair Act of 2020, Press Release. Niskanen Center, 12 August 2020. Available online: https://niskanencenter.org/niskanen-center-proud-to-endorse-criticalmedical-infrastructure-right-to-repair-act-of-2020/ (accessed on 1 August 2023).
- 113. Louviere, B. Time for a Tune Up in America's Healthcare Market: Securing the Right to Repair for Medical Devices. *J. Corp. Law* **2022**, *48*, 183–204.
- 114. Contreras, J. Patents and the Coronavirus—The Right to Repair. *InfoJustice*, 3 April 2020. Available online: http://infojustice.org/ archives/42222 (accessed on 1 August 2023).
- 115. Tur-Sinai, O.; Grinvald, L.C. Repairing Medical Equipment in Times of Pandemic. Seton Hall Law Rev. 2021, 52, 461–505. [CrossRef]
- 116. He, S.; Lai, D.; Lee, J. The Medical Right to Repair: The Right to Save Lives. Lancet 2021, 397, 1260–1261. [CrossRef]
- 117. Advanced Medical Technology Association. The "Right to Repair" is Wrong for Patients. Available online: https://www. advamed.org/our-work/key-issues/right-to-repair-wrong-for-patients/ (accessed on 1 September 2023).
- White, C. The Problems State Right-to-Repair Statutes Pose for Patients and Hospitals. Advanced Medical Technology Association. 10 June 2021. Available online: https://www.advamed.org/2021/06/10/the-problems-state-right-to-repair-statutes-pose-forpatients-and-hospitals/ (accessed on 1 September 2023).

- 119. Medical Imaging & Technology Alliance. MITA Opposes Legislation. 6 August 2020. Available online: https://www. medicalimaging.org/mita-news/view/mita-opposes-legislation-to-give-non-regulated-third-party-service-companiesaccess-to-service-tools (accessed on 1 September 2023).
- 120. Skoufalos, M. Right to Repair and Medical Imaging Device Purchasing. 1 February 2022. Available online: https://theicecommunity.com/right-to-repair-and-medical-imaging-device-purchasing/ (accessed on 1 September 2023).
- Miller, H. 'Right To Repair' Legislation Compromises Medical Device Cybersecurity. Center for Medical Economics and Innovation. 20 October 2020. Available online: https://medecon.org/right-to-repair-legislation-compromises-medical-devicecybersecurity/ (accessed on 1 September 2023).
- 122. Ghosh, S. The Continuing Right to Repair. Berkeley Technol. Law J. 2022, 37, 1097–1122.
- 123. H.R.4006—Fair Repair Act—117th Congress (2021–2022) (US). Available online: https://www.congress.gov/bill/117th-congress/ house-bill/4006 (accessed on 1 August 2023).
- 124. Hughes, M. New York Congressman Puts Forward Federal Right-to-Repair Bill: Fair Repair Act Targets all Varieties of Electronic Devices. *The Register*, 18 June 2021. Available online: https://www.theregister.com/2021/06/18/first_federal_right_to_repair_bill/ (accessed on 1 August 2023).
- 125. H.R. 6566—Freedom to Repair Act of 2022—117th Congress. Available online: https://www.congress.gov/bill/117th-congress/ house-bill/6566 (accessed on 1 September 2023).
- 126. Abbott, C. 'Right-to-Repair' Bills Languish as Time for Congressional Action Dwindles. S. 3549—Agricultural Right to Repair Act—117th Congress (2021–2022). Successful Farming, 15 September 2022. Available online: https://www.agriculture.com/news/ business/right-to-repair-bills-languish-as-time-for-congressional-action-dwindles (accessed on 1 September 2023).
- 127. Issa, D. (Rep). Issa Introduces Bipartisan Bill to Reduce Car Repair Costs. United States Congress. 22 March 2023. Available online: https://issa.house.gov/media/press-releases/issa-introduces-bipartisan-bill-reduce-car-repair-costs (accessed on 1 August 2023).
- 128. Gomulkiewicz, R.W. Considering a Right to Repair Software. Berkeley Technol. Law J. 2022, 37, 943–986.
- 129. House Judiciary Subcommittee on Courts; Intellectual Property, and the Internet. Is There a Right to Repair? In Proceedings of the 118th Congress, 2023–2024, Washington, DC, USA, 18 July 2023. Available online: https://www.congress.gov/event/118th-congress/house-event/116230?s=1&r=6 (accessed on 1 August 2023).
- 130. The White House. Executive Order on Promoting Competition in the American Economy. The White House. 9 July 2021. Available online: https://www.whitehouse.gov/briefing-room/presidential-actions/2021/07/09/executive-order-on-promoting-competition-in-the-american-economy/ (accessed on 1 August 2023).
- Federal Trade Commission. FTC To Ramp Up Law Enforcement Against Illegal Repair Restrictions, Press Release. Federal Trade Commission. 21 July 2021. Available online: https://www.ftc.gov/news-events/news/press-releases/2021/07/ftc-ramp-lawenforcement-against-illegal-repair-restrictions (accessed on 1 August 2023).
- 132. Federal Trade Commission. Remarks of Chair Lina M. Khan Regarding the Proposed Policy Statement on Right to Repair. Federal Trade Commission. 21 July 2021. Available online: https://www.ftc.gov/legal-library/browse/cases-proceedings/ public-statements/remarks-chair-lina-m-khan-regarding-proposed-policy-statement-right-repair (accessed on 1 August 2023).
- 133. SB-605 Medical Device Right to Repair Act (2021–2022) (California). Available online: https://leginfo.legislature.ca.gov/faces/ billTextClient.xhtml?bill_id=202120220SB605 (accessed on 1 August 2023).
- 134. Staff Writer. Appropriations Committee Fails to Pass California Medical Right to Repair Act, 24x7. 21 May 2021. Available online: https://24x7mag.com/standards/servicing-legislation/right-to-repair/appropriations-committee-fails-pass-california-medical-right-repair-act/ (accessed on 1 September 2023).
- World Health Organization. COVID-19: Australia Situation. 2023. Available online: https://covid19.who.int/region/wpro/ country/au (accessed on 1 August 2023).
- 136. Department of Industry, Science, Energy and Resources. 42000000—Medical Equipment and Accessories and Supplies. Australian Government. 15 March 2020. Available online: https://www.tenders.gov.au/Atm/ShowClosed/70c82ab8-bd76-4123-aa62-2c6 202d42f16?PreviewMode=False (accessed on 1 August 2023).
- 137. The Canadian Shield. Available online: https://www.canadianshieldppe.ca/ (accessed on 1 August 2023).
- 138. Zhang, K. 3D Printing Equipment for COVID-19, Pursuit. University of Melbourne. 1 May 2020. Available online: https://pursuit.unimelb.edu.au/articles/3d-printing-medical-equipment-for-covid-19 (accessed on 1 August 2023).
- 139. Johnston, M. Wollongong Uni begins 3D Printing Protective Equipment for Health Workers. *IT News*, 31 March 2020. Available online: https://www.itnews.com.au/news/wollongong-uni-begins-3d-printing-protective-equipment-for-health-workers-540030 (accessed on 1 August 2023).
- Clun, R.; Gorrey, M. 'Ready to Help': 3D-Printed Face Shields Used to Combat NSW Shortage. *The Sydney Morning Herald*, 9 April 2020. Available online: https://smh.com.au/national/nsw/ready-to-help-3d-printed-face-shields-used-to-combat-nsw-shortage-20200408-p54i5k.html (accessed on 1 August 2023).
- 141. Evlin, L. Australia's 3D Printing Experts are Banding Together to Make Masks for at-Risk Health Workers. SBS News, 3 April 2020. Available online: https://sbs.com.au/news/australia-s-3d-printing-experts-are-banding-together-to-make-masks-for-at-risk-health-workers (accessed on 1 August 2023).

- 142. University of Technology Sydney. PPE 3D Printed for PNG. University of Technology Sydney. 12 May 2020. Available online: https://www.uts.edu.au/about/faculty-engineering-and-information-technology/news/ppe-3d-printed-png (accessed on 1 August 2023).
- ANU. Senior Lecturer 3D Prints Face Shields for GPs. ANU News, 21 April 2020. Available online: https://soad.cass.anu.edu.au/ news/anu-senior-lecturer-3d-prints-face-shields-gps (accessed on 1 August 2023).
- 144. ANU. ANU helps make Protective Equipment for Health Workers. ANU News, 1 May 2020. Available online: https://www.anu.edu.au/ news/all-news/anu-helps-make-protective-equipment-for-health-workers (accessed on 1 August 2023).
- ANU. ANU MakerSpace Leads Push to Make Free Protective Equipment for Health Workers. ANU News, 12 May 2020. Available online: https://physics.anu.edu.au/news_events/?NewsID=193 (accessed on 1 August 2023).
- 146. CSIRO. Face Shields: We're using 3D Printing to Produce Two Designs of Protective Face Shields for Queensland Healthcare Workers. CSIRO, 2 March 2021. Available online: https://www.csiro.au/en/research/health-medical/diseases/covid-19-research/face-shields (accessed on 1 August 2023).
- 147. Free 3D Hands. Available online: https://free3dhands.org/ (accessed on 1 August 2023).
- 148. Monash Life. 3D-Printed Prosthetic Limbs: How a Social Entrepreneur is Changing Lives. *Monash Life*, 2019. Available online: https://www.monash.edu/alumni/community/monash-life/2019/mat-bowtell-printed-prosthetic-limbs (accessed on 1 August 2023).
- 149. Osman, H. Aussie 3D Printing Company Produces COVID-19 Swabs. *Sprinter*, 6 August 2020. Available online: https://www.sprinter.com.au/3d-printing-studios-becomes-the-first-local-supplier-of-these-essential-products-endingaustralian-reliance-on-overseas-supply/ (accessed on 1 August 2023).
- 150. Wide Format Online, Australian 3D Printers Join Global Program to supply Face Shields and Ventilator Masks. *Wide Format Online*. Available online: https://www.wideformatonline.com/news/wide-format-news/10418-australian-3d-printers-join-global-program-to-supply-face-shields-ventilator-masks.html (accessed on 1 August 2023).
- 151. Kanapathippillai, J. 3D Printing Hobbyists Make Protective Gear for Canberra Hospitals. *The Canberra Times*, 13 April 2020. Available online: https://canberratimes.com.au/story/6716923/3d-printing-hobbyists-make-protective-gear-for-canberra-hospitals/?cs=17267 (accessed on 1 August 2023).
- 152. Rimmer, M. 'Shane Rattenbury, the Productivity Commission, and the Right to Repair: Intellectual Property, Consumer Rights, and Sustainable Development in Australia'. *Berkeley Technol. Law J.* **2022**, *37*, 989–1056.
- 153. Productivity Commission. *Right To Repair: Issues Paper*; Productivity Commission: Melbourne, Australia, 2020. Available online: https://www.pc.gov.au/inquiries/completed/repair/issues (accessed on 1 August 2023).
- 154. Productivity Commission. *Right to Repair: Draft Report;* Productivity Commission: Melbourne, Australia, 2021. Available online: https://www.pc.gov.au/inquiries/current/repair/draft (accessed on 1 August 2023).
- 155. Productivity Commission. *Right to Repair: Inquiry Report No. 97*; Productivity Commission: Melbourne, Australia, 2021. Available online: https://www.pc.gov.au/inquiries/completed/repair/report (accessed on 1 August 2023).
- 156. Rimmer, M. *A Submission to the Productivity Commission Inquiry on the Right to Repair;* Productivity Commission: Melbourne, Australia, 2021. Available online: https://eprints.qut.edu.au/212034/ (accessed on 1 August 2023).
- 157. Abbas, M.Z. Patent Law and 3D Printing Applications in Repairing Medical Equipment, Submission to the Productivity Commission Inquiry into the Right to Repair in Australia. Productivity Commission. January 2021. Available online: https://www.pc.gov.au/__data/assets/pdf_file/0006/272292/sub034-repair.pdf (accessed on 1 August 2023).
- 158. Abbas, M.Z. Patent Law and 3D Printing Applications in Response to COVID-19: Exceptions to Inventor Rights. *J. World Intellect. Prop.* **2022**, *25*, 317–334. [CrossRef]
- 159. Medtronic Australia. *Productivity Commission: Inquiry into the Right to Repair*; Submission 186 to the Productivity Commission Inquiry into the Right to Repair in Australia; Productivity Commission: Melbourne, Australia, 2021. Available online: https://www.pc.gov.au/__data/assets/pdf_file/0004/279220/subdr186-repair.pdf (accessed on 1 August 2023).
- 160. Productivity Commission. *Vulnerable Supply Chains*; Productivity Commission: Melbourne, Australia, 2021; p. 1. Available online: https://www.pc.gov.au/inquiries/completed/supply-chains/report (accessed on 1 August 2023).
- Leigh, A. Keynote Address to the Australian Repair Summit 2022, Canberra. Australian Government, 5 August 2022. Available online: https://ministers.treasury.gov.au/ministers/andrew-leigh-2022/speeches/keynote-address-australian-repair-summit-2022-canberra (accessed on 1 September 2023).
- Leigh, A. Australian Government Driving Down The Cost of Car Repairs. Australian Government, 30 June 2022. Available online: https://ministers.treasury.gov.au/ministers/andrew-leigh-2022/media-releases/australian-government-driving-downcost-car-repairs (accessed on 1 September 2023).
- 163. Powell, M.; Brown, A.; Long, W. Right to Repair Changes Floated as Frustrated Farmers Count Cost of Harvest Delays. *ABC Rural*, 13 March 2023. Available online: https://www.abc.net.au/news/rural/2023-03-13/right-to-repair-on-horizon-as-governmentpromises-farmers-action/102078970 (accessed on 1 September 2023).
- 164. Sarnoff, J. TRIPS, COVID-19 and the Right to Repair and Produce Needed Medical Products in Emergencies. *TradeRX*, 11 June 2020. Available online: https://traderxreport.com/covid-19/trips-covid-19-and-the-right-to-repair-and-produce-needed-medical-products-in-emergencies-part-2-of-2/ (accessed on 1 August 2023).
- Doha Declaration on the TRIPS Agreement and Public Health 2001. In Proceedings of the WT/MIN(01)/DEC/2 Ministerial Conference, Fourth Session, Doha, Qatar, 9–14 November 2001.

- 166. Abbott, F. The Doha Declaration on the TRIPS Agreement and Public Health: Lighting a Dark Corner at the WTO. *J. Int. Econ. Law* **2022**, *5*, 469–505. [CrossRef]
- 167. WTO. Implementation of Paragraph 6 of the Doha Declaration on the TRIPS Agreement and Public Health; WTO Doc WT/L/540; (WTO General Council Decision 2003).
- 168. Flynn, S.; Nkrumah, E.; Schirru, L. Non-Patent Intellectual Property Barriers to COVID-19 Vaccines, Treatment and Containment. Digital Commons. American University Washington College of Law. November 2021. Available online: https://digitalcommons. wcl.american.edu/research/71/ (accessed on 1 August 2023).
- MSF. US Should Stop Blocking COVID-19 WTO Waiver to Boost Vaccines, Treatments Worldwide, Press Release. MSF, 26 February 2021. Available online: https://www.doctorswithoutborders.org/latest/us-should-stop-blocking-covid-19-wto-waiver-boostvaccines-treatments-worldwide (accessed on 23 February 2023).
- 170. United States Trade Representative. Statement from Ambassador Katherine Tai on the COVID-19 TRIPS Waiver, Press Release. United States Trade Representative. 5 May 2021. Available online: https://ustr.gov/about-us/policy-offices/press-office/ press-releases/2021/may/statement-ambassador-katherine-tai-covid-19-trips-waiver (accessed on 1 August 2023).
- 171. Tehan, D. Press Conference: TRIPS Waiver, Department of Foreign Affairs and Trade. Australian Government. 8 September 2021. Available online: https://www.trademinister.gov.au/minister/dan-tehan/transcript/press-conference-1 (accessed on 1 August 2023).
- O'Connor, D. NZ Backs Moves To Improve Global Access to COVID Vaccines, Press Release. New Zealand Government. 6 May 2021. Available online: https://www.beehive.govt.nz/release/nz-backs-moves-improve-global-access-covid-vaccines (accessed on 1 August 2023).
- 173. Karp, P.; Visontay, E. Australia to Support Vaccine Waiver after Months of Pressure from Human Rights Groups. *The Guardian*, 8 September 2021. Available online: https://www.theguardian.com/society/2021/sep/08/australia-to-support-vaccine-waiverafter-months-of-pressure-from-human-rights-groups (accessed on 1 August 2023).
- 174. Corporate Europe. TRIPS "Waiver Failure": EU Betrayal of Global South on Vaccine Access Obscured by Lack of Transparency. *Corporate Europe*, 8 July 2022. Available online: https://corporateeurope.org/en/2022/07/trips-waiver-failure-eu-betrayal-global-south-vaccine-access-obscured-lack-transparency (accessed on 1 August 2023).
- 175. Human Rights Watch. Seven Reasons the EU is Wrong to Oppose the TRIPS Waiver. *Human Rights Watch*, 3 June 2021. Available online: https://www.hrw.org/news/2021/06/03/seven-reasons-eu-wrong-oppose-trips-waiver (accessed on 1 August 2023).
- European Union. Urgent Trade Responses to the COVID-19 Crisis: Intellectual Property, Communication from the European Union to the Council for TRIPS. 4 June 2021. Available online: https://ec.europa.eu/commission/presscorner/detail/en/IP_21_2801 (accessed on 1 August 2023).
- 177. MSF. 3 Reasons why the EU's TRIPS Waiver Counter-Proposal is an Insufficient Solution in a Pandemic. *MSF Access*, 18 June 2021. Available online: https://msfaccess.org/3-reasons-why-eus-trips-waiver-counter-proposal-insufficient-solution-pandemic (accessed on 1 August 2023).
- 178. European Union; India; South Africa; United States. Quad Proposal on TRIPS COVID-19; IP/C/W/688.
- World Trade Organization. Quad's Outcome Document on IP COVID-19 Response Made Public, Press Release. World Trade Organization. 3 May 2022. Available online: https://www.wto.org/english/news_e/news22_e/trip_03may22_e.htm (accessed on 1 August 2023).
- World Trade Organization. Members to Stretch Deadline on Extending TRIPS Decision to COVID Diagnostics, Therapeutics, Press Release. World Trade Organization. 16 December 2022. Available online: https://www.wto.org/english/news_e/news22_ e/trip_15dec22_e.htm (accessed on 1 August 2023).
- 181. United States Trade Representative. Statement from Ambassador Katherine Tai on an Intellectual Property Response to the COVID-19 Pandemic, Press Release. United States Trade Representative. 17 June 2022. Available online: https: //ustr.gov/about-us/policy-offices/press-office/press-releases/2022/june/statement-ambassador-katherine-tai-intellectualproperty-response-covid-19-pandemic (accessed on 1 August 2023).
- Cullinan, K. US Wants WTO To Delay Decision on TRIPS Waiver Extension for COVID-19 Tests and Treatments. *Health Policy Watch*, 6 December 2022. Available online: https://healthpolicy-watch.news/us-wants-wto-to-delay-decision-on-trips-waiver-extension-for-covid-19-tests-and-treatment/ (accessed on 1 August 2023).
- 183. United States International Trade Commission. USITC To Report on COVID-19 Diagnostics and Therapeutics and Flexibilities under the TRIPS Agreement, Press Release. United States International Trade Commission. 1 February 2023. Available online: https://www.usitc.gov/press_room/news_release/2023/er0201_63483.htm (accessed on 1 August 2023).
- 184. Farrell, D.; Ayres, T. A Boost in Confidence at the World Trade Organization, Press Release. Department of Foreign Affairs and Trade, Australian Government. 19 June 2022. Available online: https://ministers.dfat.gov.au/minister/tim-ayres/media-release/ boost-confidence-world-trade-organization (accessed on 1 August 2023).
- Choong, Y.Y.C.; Tan, H.W.; Patel, D.C.; Choong, W.T.N.; Chen, C.-H.; Low, H.Y.; Tan, M.J.; Patel, C.D.; Chua, C.K. The Global Rise of 3D Printing During the COVID-19 Pandemic. *Nat. Rev. Mater.* 2020, *5*, 637–639. [CrossRef] [PubMed]
- 186. United Nations Development Programme. 21st Century Development: 60 Accelerator Labs Aim to 'Future Proof' Progress. United Nations Development Programme. 18 July 2019. Available online: https://www.undp.org/press-releases/21st-centurydevelopment-60-accelerator-labs-aim-future-proof-progress (accessed on 1 August 2023).

- United Nations General Assembly. Transforming Our World: The 2030 Agenda for Sustainable Development; A/RES/70/1; United Nations General Assembly: New York, NY, USA. Available online: https://sdgs.un.org/2030agenda (accessed on 1 August 2023).
- Bachelet, M. A Joint Appeal for Open Science by CERN, OHCHR, UNESCO, and WHO, Office of the Higher Commissioner for Human Rights, United Nations. Available online: https://www.ohchr.org/en/press-releases/2020/10/joint-appeal-openscience-cern-ohchr-unesco-and-who (accessed on 27 October 2020).
- The UNESCO Recommendation on Open Science. 2021. Available online: https://en.unesco.org/science-sustainable-future/ open-science/recommendation (accessed on 1 August 2023).
- 190. Strandburg, K.; Frischmann, B.; Madison, M. (Eds.) *Governing Medical Knowledge Commons;* Cambridge University Press: Cambridge, UK, 2017.
- 191. Molitch, M. COVID-19 and 3D Printing: Go Open Source to Save Lives. *3DPrint.com*, 13 May 2020. Available online: https://3dprint.com/267219/covid-19-and-3d-printing-go-open-source-to-save-lives/ (accessed on 1 August 2023).
- 192. Quicker, K.; Holmes, L.; Rosebush, L.; Wagner, M. Considerations for 3D Printing of Medical Devices, Accessories, Components and Parts During the COVID—19 Pandemic. *IP Intelligence Report*, 13 April 2020. Available online: https: //www.ipintelligencereport.com/2020/04/13/considerations-for-3d-printing-of-medical-devices-accessories-componentsand-parts-during-the-covid-19-pandemic/ (accessed on 1 August 2023).
- 193. Daly, A. Medical 3D Printing, Intellectual Property and Regulation. In *3D Printing in Medicine*, 2nd ed.; Kalaskar, D., Ed.; Woodhead Publishing: Cambridge, MA, USA; Elsevier: Amsterdam, The Netherlands, 2023; pp. 385–398.
- European Commission. Conformity Assessment Procedures for 3D Printing and 3D Printing Products to Be Used in the Medical Context for COVID-19. 31 March 2020. Available online: https://ec.europa.eu/docsroom/documents/40562 (accessed on 1 August 2023).
- 195. United States Food and Drug Administration. 3D Printing of Medical Devices, Accessories, Components, and Parts during the COVID-19 Pandemic. 2020. Available online: https://web.archive.org/web/20230414155056/https://www.fda.gov/medical-devices/ coronavirus-covid-19-and-medical-devices/3d-printing-medical-devices-accessories-components-and-parts-during-covid-19 -pandemic (accessed on 1 August 2023).
- Therapeutic Goods Administration. Manufacturing Medical Devices for COVID-19 including 3-D Printing, Therapeutic Goods Administration. 28 January 2021. Available online: https://www.tga.gov.au/resources/resource/guidance/manufacturingmedical-devices-covid-19-including-3-d-printing (accessed on 1 August 2023).
- 197. Engstrom, N.F. 3D Printing and Product Liability: Identifying the Obstacles. Univ. Pa. Law Rev. 2013, 162, 35-41.
- 198. Berkowitz, N. Strict Liability for Individuals? The Impact of 3D Printing on Products Liability Law. *Wash. Univ. Law Rev.* 2015, 92, 1019–1053.
- Lindenfeld, E. 3D Printing of Medical Devices: CAD Designers as the Most Realistic Target for Strict, Product Liability Lawsuits. Univ. Mo.-Kans. City Law Rev. 2016, 85, 79–103. [CrossRef]
- Higgins, C. COVID19 IP Update: Can 3D Printing Be the Solution to the Medical Supply Shortage? Orrick, 22 March 2020. Available online: https://www.orrick.com/en/Insights/2020/03/COVID19-IP-Update-Can-3D-Printing-Be-the-Solution-to-the-Medical-Supply-Shortage (accessed on 1 August 2023).
- Vrasmasu, M. COVID-19: Will 3-D Printing Medical Supplies and Protective Products Expose You to Liability? *JDSupra*, 9 April 2020. Available online: https://www.jdsupra.com/legalnews/covid-19-will-3-d-printing-medical-11248/ (accessed on 1 August 2023).
- Edwards, M.; Zabbo, T.; Hochberg, M. 3D Printing: New Frontier in Potential Product Liability Claims. New England Biz Law, 14 February 2023. Available online: https://newenglandbizlawupdate.com/2023/02/14/3d-printing-new-frontier-in-potentialproduct-liability-claims/ (accessed on 1 August 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.